		INDEX	
		GENERAL RULES AND REGULATIONS FOR ELECTRIC SERVICE	
			Sheet No.
Int	roducti		6.010
1 1		ce Agreements	6.010
1	1.1	Application for Service	6.010
	1.2	Information needed	6.010
	1.2	Agreement	6.010
	1.4	Applications by Agents	6.010
	1.5	Prior Indebtedness	6.010
	1.6	Discontinuance of Service	6.010
	1.65	Medically Essential Service	6.011
	1.7	Reimbursement for Extra Expenses	6.011
2	Supp	ly and Use of Service	6.011
	2.1	Service	6.011
	2.2	Availability of Service	6.020
	2.3	Point of Delivery	6.020
	2.4	Character of Service	6.020
	2.5	Continuity of Service	6.020
	2.6	Temporary Service	6.020
	2.7	Indemnity to Company	6.020
	2.71	Indemnity to Company – Governmental	6.020
	2.8	Access to Premises	6.020
	2.9	Right of Way	6.020
3	Limit	ation of Use	6.020
	3.1	Resale of Service Prohibited	6.020
	3.2	Street Crossings	6.030
	3.3	Unauthorized Use of Service	6.030
	3.4	Conversion to Master Metering Prohibited	6.030
4	Custo	omer's Installation	6.030
	4.1	Customer's Installation	6.030
	4.2	Type and Maintenance	6.030
	4.3	Change of Customer's Installation	6.030
	4.4	Inspection of Customer's Installation	6.030
	4.5	Electric Generators	6.030
	4.6	Momentary Parallel Operation	6.030
5	Com	pany's Installation	6.040
	5.1	Protection of Company's Property	6.040
	5.2	Damage to Company's Property	6.040
	5.3	Relocation of Company's Facilities	6.040
	5.4	Attachments to Poles	6.040
	5.5	Interference with Company's Facilities	6.040
	5.6	Unobstructed Access to Company's Facilities	6.040
6		ity Deposits/Guaranties	6.040
	6.1	Security Deposit/Guaranty	6.040
	6.2	Deposit Interest	6.050
	6.21	Residential Deposits	6.050
	6.22	Non-residential Deposits	6.050
	6.3	Refund of Cash Deposit/Release of other Security or Guaranty	6.050
	6.4	Transfer of Security Deposit/Guaranty	6.050

INDEX

(Continued on Sheet No. 6.002)

(Continued from Sheet No. 6.001)

GENERAL RULES AND REGULATIONS FOR ELECTRIC SERVICE

7	Billing		6.050
	7.1	Billing Periods	6.050
	7.11	Regular Bills	6.050
	7.12	Prorated Bill	6.050
	7.13	Month	6.050
	7.14	Budget Billing	6.052
	7.14.1	Residential	6.052
	7.14.2	Non-residential	6.052
	7.15	Summary Billing	6.060
	7.2	Non-Receipt of Bills	6.060
	7.3	Evidence of Consumption	6.060
	7.4	Application of Rate Schedules	6.060
	7.5	Optional Rate	6.060
	7.6	Taxes and Charges	6.060
	7.7	Disconnection and Reconnection of Service	6.060
	7.71	Disconnection of Service	6.060
	7.72	Reconnection of Service	6.060
	7.8	Change of Occupancy	6.060
	7.9	Delinquent Bills	6.060
8	Meters		6.061
	8.1	Location of Meters	6.061
	8.2	Setting and Removing Meters	6.061
	8.3	Tampering with Meters	6.061
	8.4	Meter Tests	6.061
	8.5	Failure of Meter	6.061
9	Service	Standards	6.061
10	Installat	ion of Underground Electric Distribution Facilities	
		ve Residential Customers	6.080
	10.1	Definitions	6.080
	10.2	General	6.090
	10.2.1	Application	6.090
	10.2.2	Early Notification and Coordination	6.090
	10.2.3	Changes to Plans, Layout or Grade	6.090
	10.2.4	Underground Installations Not Covered	6.090
	10.2.5	Type of System Provided	6.090
	10.2.6	Design and Ownership	6.090
	10.2.7	Rights of Way and Easements	6.090
	10.2.8	Contributions and Credits	6.090
	10.2.8.1	Credit for TUGs	6.095
	10.2.9	Location of Distribution Facilities	6.095
	10.2.10	Special Conditions	6.095
	10.2.11	Point of Delivery	6.095
	10.2.12	Location of Meter and Downpipe	6.096
	10.2.13	Relocation or Removal of Existing Facilities	6.096
	10.2.14	Development of Subdivisions	6.096

(Continued on Sheet No. 6.003)

11

(Continued from Sheet No. 6.002)

GENERAL RULES AND REGULATIONS FOR ELECTRIC SERVICE

			Sheet No.
10.3	Undergro	ound Distribution Facilities for Residential	
	-	isions and Developments	6.100
	10.3.1	Availability	6.100
	10.3.2	Contribution by Applicant	6.100
	10.3.3	Contribution Adjustments	6.110
10.4		nd Service Laterals from Overhead	
	Electric D	Distribution Systems	6.120
	10.4.1	New Underground Service Laterals	6.120
	10.4.2	Contribution by Applicant	6.120
	10.4.3	Contribution Adjustments	6.120
10.5	Undergroun	d Service Laterals Replacing Existing Residential	
	Overhead	Services and Underground Services	6.130
	10.5.1	Applicability	6.130
	10.5.2	Rearrangement of Service Entrance	6.130
	10.5.3	Trenching and Conduit Installation	6.130
	10.5.4	Contribution by Applicant	6.130
10.6		d Service Distribution Facilities to Multiple-Occupancy	
	Residentia	al Buildings	6.140
	10.6.1	Availability	6.140
	10.6.2	Contribution by Applicant	6.140
	10.6.3	Responsibility of Applicant	6.140
	10.6.4	Responsibility of the Company	6.140
	10.6.5	Service Voltages	6.150
Instal	lation of Nev	v or Upgraded Facilities	6.199
11.1	General		6.199
11.1.1	Contribu	tion-In-Aid of Construction (CIAC)	6.199
11.1.2	CIAC Tr	ue-Up	6.199
11.1.3		of CIAC	6.200
11.2	Installatio	on of Underground Electric Distribution Facilities for New Construction	6.200
11.2.0		ion System	6.200
11.2.1	Applicati		6.200
11.2.2		tion-in-Aid-of-Construction (CIAC)	6.200
11.2.3		undable Deposits	6.200
11.2.4		ding Cost Estimates	6.210
11.2.5	-	ound Distribution Facilities Installation Agreement	6.210
11.2.6			6.210
11.2.7		tification and Coordination	6.210
11.2.8		to Plans, Layout or Grade	6.210
11.2.9		of Distribution Facilities	6.210
11.2.1		rms and Conditions	6.210
11.2.1	• •	System Provided	6.220
11.2.1	2 Design a	nd Ownership	6.220

(Continued on Sheet No. 6.004)

(Continued from Sheet No. 6.003)

GENERAL RULES AND REGULATIONS FOR ELECTRIC SERVICE

Sheet No.

12	Install	ation of Un	nderground Electric Distribution Facilities for the	
	Con	version of (Overhead Electric Distribution Facilities	6.300
	12.1	Definition	ns	6.300
	12.2	General		6.301
		12.2.1	Application	6.301
		12.2.2	Contribution-in-Aid-of-Construction (CIAC)	6.301
		12.2.3	Non-Refundable Deposits	6.310
		12.2.4	Non-Binding Cost Estimates	6.310
		12.2.5	Underground Facilities Conversion Agreement	6.310
		12.2.6	Simultaneous Conversion of Other Pole Licenses	6.310
		12.2.7	Easements	6.310
		12.2.8	Affected Customer Services	6.320
		12.2.9	Other Terms and Conditions	6.320
		12.2.10	Type of System Provided	6.330
		12.2.11	Design and Ownership	6.330
		12.2.12	Relocation	6.330
13	Supple	ement to Ge	eneral Rules and Regulations for the Installation of	
10	Underground Electric Distribution Facilities to Serve			
		0	cial/Industrial Customers	6.400
	13.1	Definition		6.400
	13.2	Underground Distribution Facilities to Small		
		Commercial/Industrial Customers		6.500
		13.2.1	Application	6.500
		13.2.2	Early Notification and Coordination	6.500
		13.2.3	Changes to Plans, Layout or Grade	6.500
		13.2.4	Type of System Provided	6.500
		13.2.5	Design and Ownership	6.500
		13.2.6	Rights of Way and Easements	6.510
		13.2.7	Contribution and Credits	6.510
		13.2.8	Location of Distribution Facilities	6.510
		13.2.9	Special Conditions	6.510
		13.2.10	Point of Delivery	6.510
		13.2.11	Location of Meter and Raceway	6.510
		13.2.12	Contribution by Applicant	6.520
		13.2.13	Contribution Adjustments	6.540

GENERAL RULES AND REGULATIONS FOR ELECTRIC SERVICE

INTRODUCTION

These General Rules and Regulations are a part of the Company's Tariff, covering the terms and conditions under which Electric Service is supplied by the Company to the Customer. They are supplementary to the "Rules and Regulations Governing Electric Service by Electric Utilities" issued by the Florida Public Service Commission.

1 SERVICE AGREEMENTS

<u>1.1</u> <u>Application for Service</u>. Service may be obtained upon application. Usually all that is required is the service application, a form of identification acceptable to the Company, and the posting of a guarantee deposit.

<u>1.2 Information Needed</u>. To provide service promptly the Company will need the applicant's name, telephone number and address including the street, house number (or apartment number), or the name of the subdivision with lot and block numbers. The types of identification required upon application for service include a valid social security number, tax identification number, driver's license, birth certificate or any other form of identification acceptable to the Company. On new or changed installations, the Company will also need to know the equipment that will be used. The Company will advise the Customer as to whether the desired type of service is available at the designated location.

<u>1.3 Agreement</u>. Service is furnished upon acceptance of the agreement or contract by the Company. Applications are accepted by the Company with the understanding that there is no obligation to render service other than the character of service then available at the point of delivery. A copy of any written agreement accepted by the Company will be furnished to the applicant upon request.

<u>1.4 Applications by Agents</u>. Applications for service requested by firms, partnerships, associations, corporations, etc., shall be made only by duly authorized parties. When service is rendered under an agreement or agreements entered into between the Company and an agent of a principal, the use of such service by the principal shall constitute full and complete ratification by the principal of such agreement or agreements.

<u>1.5</u> Prior Indebtedness. The Company may refuse or discontinue service for failure to settle, in full, all prior indebtedness incurred by any Customer(s) for the same class of service at any one or more locations of such Customer(s). The Company may also refuse service for prior indebtedness by a previous customer provided that the current applicant or customer occupied the premises at the time the prior indebtedness occurred and the previous customer continues to occupy the premises.

<u>1.6 Discontinuance of Service</u>. Service may be discontinued for violation of the Company's rules or by actions or threats made by a customer, or anyone on the customer's premises, which are reasonably perceived by a utility employee as violent or unsafe, after affording the Customer reasonable opportunity to comply with said rules, and/or the customer agrees to cease from any further act of violence or unsafe condition, including five (5) days written notice to the Customer. However, where the Company believes a dangerous condition exists on the Customer's premises, service may be discontinued without notice.

(Continued on Sheet No. 6.011)

(Continued from Sheet No. 6.010)

<u>1.65 Medically Essential Service</u>. For purposes of this section, a Medically Essential Service Customer is a residential customer whose electric service is medically essential, as affirmed through the certificate of a doctor of medicine licensed to practice in the State of Florida. Service is "medically essential" if the customer has a medical dependence on electric-powered equipment that must be operated continuously or as circumstances require as specified by a physician to avoid the loss of life or immediate hospitalization of the customer or another permanent resident at the residential service address. If continuously operating, such equipment shall include but is not limited to the following: oxygen concentrator or a ventilator/respirator. The physician's certificate shall explain briefly and clearly, in non-medical terms, why continuance of service is medically essential, and shall be in the form of tariff sheet no. 9.930. A customer who is certified as a Medically Essential Service Customer must renew such certification periodically through the procedures outlined above. The Company may require such renewed certification no more frequently than once every 12 months.

The Company shall provide Medically Essential Service Customers with a limited extension of time, not to exceed thirty (30) days, beyond the date service would normally be subject to disconnection for non-payment of bills (following the requisite notice pursuant to Rule 25-6.105(5) of the Florida Administrative Code). The Company shall provide the Medically Essential Service Customer with written notice specifying the date of disconnection based on the limited extension. The Medically Essential Service Customer shall be responsible for making mutually satisfactory arrangements to ensure payment within this additional extension of time for services provided by the Company and for which payment is past due, or to make other arrangements for meeting the medically essential needs.

No later than 12 noon one day prior to the scheduled disconnection of service of a Medically Essential Service Customer, the Company shall attempt to contact such customer by telephone in order to provide notice of the scheduled disconnect date. If the Medically Essential Service Customer does not have a telephone number listed on the account, or if the utility cannot reach such customer or other adult resident of the premises by telephone by the specified time, a field representative will be sent to the residence to attempt to contact the Medically Essential Service Customer, no later than 4 PM of the day prior to scheduled disconnection. If contact is not made, however, the Company may leave written notification at the residence advising the Medically Essential Service Customer of the scheduled disconnect date; thereafter, the Company may disconnect service on the specified date. The Company will grant special consideration to a Medically Essential Service Customer in the application of Rule 25-6.097(3) of the Florida Administrative Code.

In the event that a customer is certified as a Medically Essential Service Customer, the customer shall remain solely responsible for any backup equipment and/or power supply and a planned course of action in the event of power outages. The Company does not assume, and expressly disclaims, any obligation or duty: to monitor the health or condition of the person requiring medically essential service; to insure continuous service; to call, contact, or otherwise advise of service interruptions; or, except as expressly provided by this section, to take any other action (or refrain from any action) that differs from the normal operations of the Company.

<u>1.7 Reimbursement for Extra Expenses</u>. The Customer may be required to reimburse the Company for all extra expenses incurred by the Company on account of violations by the Customer of agreements with the Company or the Rules and Regulations of the Company.

2 SUPPLY AND USE OF SERVICE

2.1 Service. Service includes all power and energy required by the Customer and, in addition, the readiness and ability on the part of the Company to furnish power and energy to the Customer. Thus, the maintenance by the company of approximately the agreed voltage and frequency at the point of delivery shall constitute the rendering of service, irrespective of whether the Customer makes any use thereof.

(Continued on Sheet No. 6.020)

2.2 Availability of Service. The Company will supply electric service to any applicant for service throughout the territory it serves, subject to the following conditions: should an extension of the Company's facilities be required, the Company will pay for the cost where justified, in the Company's opinion, by revenues to be secured; however, the Company may require monthly or annual guarantees, cash contributions in aid of construction, and/or advances for construction, when in the Company's opinion, the immediate or potential revenues do not justify the cost of extension. If facilities are requested that are not usual and customary for the type of installation to be served, the Company may require a contribution in aid of construction based upon the incremental cost of the requested facility. All contributions in aid of construction will be calculated in accordance with applicable rules and regulations of the Florida Public Service Commission. If the installation of facilities is justified based on the Customer's estimates for electric power but there is reasonable doubt as to level of use or length of use of such facilities, the Customer, when mutually agreeable with the Company, may contract for a minimum Demand or monthly payment sufficient to justify the Company's investment. Upon request, written information will be supplied by the Company concerning the availability and character of service for any desired location. The Company will not be responsible for mistakes of any kind resulting from information given orally.

2.3 Point of Delivery. This is the point where the Company's wires or apparatus are connected with those of the Customer. The point of delivery shall be determined by the Company.

2.4 Character of Service. Alternating current is supplied at a frequency of approximately sixty cycles. Standard nominal voltages are 120 or 120/240 volts for single-phase service and 240 volts for 3-phase delta service. Where three-phase "Wye" service is provided, the standard nominal voltages are 120/208 or 277/480 volts. In some locations other voltages are available. The Company will furnish information regarding Character of Service on request.

2.5 Continuity of Service. The Company will use reasonable diligence at all times to provide continuous service at the agreed nominal voltage, and shall not be liable to the Customer for complete or partial failure or interruption of service, or for fluctuations in voltage, resulting from causes beyond its control or through the ordinary negligence of its employees, servants or agents. The Company shall not be liable for any act or omission caused directly or indirectly by strikes, labor troubles, accident, litigation, shutdowns for repairs or adjustments, interference by Federal, State or Municipal governments, acts of God or other causes beyond its control.

<u>2.6 Temporary Service</u>. Temporary service refers to service required for short-term exhibitions, displays, bazaars, fairs, construction work, houseboats, dredging jobs, and the like. It will be supplied only when the Company has readily available capacity of lines, transformers, generating and other equipment for the service requested. Before supplying temporary service the Company may require the Customer to bear the cost of installing and removing the necessary service facilities, less credit for salvage.

2.7 Indemnity to Company. The Customer shall indemnify, hold harmless and defend the Company from and against any and all liability, proceedings, suits, cost or expense for loss, damage or injury to persons or property, in any manner directly or indirectly connected with, or growing out of the transmission and use of electricity on the Customer's side of the point of delivery.

2.71 Indemnity to Company - Governmental. Notwithstanding anything to the contrary in the Company's tariff, including these General Rules and Regulations for Electric Service, the Company's Rate Schedules, and its Standard Forms, any obligation of indemnification therein required of a Customer, Applicant, or QF, that is a governmental entity of the State of Florida or political subdivision thereof ("governmental entity"), shall be read to include the condition "to the extent permitted by applicable law."

2.8 Access to Premises. The duly authorized agents of the Company shall have safe access to the premises of the Customer at all reasonable hours for the purpose of installing, maintaining, and inspecting or removing the Company's property, reading meters, trimming trees within the Company's easements and rights of way, and other purposes incident to performance under or termination of the Company's agreement with the Customer, and in such performance shall not be liable for trespass.

2.9 Right of Way. The Customer shall grant or cause to be granted to the Company and without cost to the Company all rights, easements, permits and privileges which, in the opinion of the Company, are necessary for the rendering of service to the Customer.

3 LIMITATION OF USE

3.1 Resale of Service Prohibited. Electric service received from the Company shall be for the Customer's own use and shall not be resold. Where individual metering is not required under Subsection (5) of Section 25-6.049 (Measuring Customer Service) of the Florida Administrative Code and master metering is used in lieu thereof, reasonable apportionment methods, including sub-metering, may be used by the Customer solely for the purpose of allocating the cost of the electricity billed by the utility. Any fees or charges collected by a Customer for electricity billed to the Customer's account by the utility, whether based on the use of sub-metering or any other allocation method, shall be determined in a manner which reimburses the Customer for no more than the Customer's actual cost of electricity.

For the purpose of this Rule:

- (1) Electric service is "sub-metered" when separate electric meters are used to allocate among tenants, lessees or other entities the monthly bill rendered by FPL to the Customer for electric service, when these tenants, lessees or other entities are charged no more than a proportionate share of such bill, based on their monthly consumption as measured by such meters.
- (2) Electric service is "resold" when separate electric meters are used to charge tenants, lessees or other entities more than a proportionate share of the Customer's monthly bill.
- (3) The term "cost" as used herein means only those charges specifically authorized by FPL's tariff, including but not limited to the customer, energy, demand, fuel, conservation, capacity and environmental charges plus applicable taxes and fees to the customer of record responsible for the master meter payments. The term does not include late payment charges, returned check charges, the cost of the customer-owned distribution system behind the master meter, the customer of record's cost of billing the individual units, and other such costs.

<u>3.2 Street Crossings</u>. The Customers may not build or extend his/her lines across or under a street, alley, lane, court, avenue or other way in order to furnish service for adjacent property through one meter even through such adjacent property is owned by the Customer, unless written consent is obtained from the Company. Consent may be given when such adjacent properties are operated as one integral unit, under the same name, for carrying on parts of the same business.

<u>3.3 Unauthorized Use of Service</u>. In case of any unauthorized remetering, sale, extension or other disposition of service, the Customer's service is subject to discontinuance until such unauthorized remetering, sale, extension or other disposition of service is discontinued, full payment is made of bills for service calculated on proper classifications and rate schedules, and reimbursement in full has been made to the Company for all extra expenses incurred, including expenses for clerical work, testing and inspections.

<u>3.4 Conversion to Master Metering Prohibited</u>. When customers are currently separately served by the Company as individual accounts, they may not terminate these individual accounts and receive service from the Company collectively through a single meter account unless the resulting combined service account is one which could be served by one meter in accordance with Rule 25-6.049 Section (5) of the Florida Administrative Code.

4 CUSTOMER'S INSTALLATION

4.1 Customer's Installation. The Customer's installation consists of and includes all wires, cutouts, switches and appliances and apparatus of every kind and nature used in connection with or forming a part of an installation for utilizing electric service for any purpose, (excepting meters and associated equipment), ordinarily located on the Customer's side of "Point of Delivery," and including "Service Entrance Conductors," whether such installation is owned outright by the Customer or used by the Customer under lease or otherwise.

<u>4.2 Type and Maintenance</u>. The Customer's wires, apparatus and equipment shall be selected and used with a view to obtaining the highest practicable power factor, and shall be installed and maintained in accordance with standard practice, and in full compliance with all applicable laws, codes and governmental and Company regulations. The Customer expressly agrees to utilize no apparatus or device which is not properly constructed, controlled and protected, or which may adversely affect service to others, and the Company reserves the right to discontinue or withhold service for such apparatus or device.

<u>4.3 Change of Customer's Installation</u>. No changes or increases in the Customer's installation, which will materially affect the operation of any portion of the distribution system or generating plants of the Company shall be made without written consent of the Company. The Customer will be liable for any damage resulting from a violation of this rule.

<u>4.4 Inspection of Customer's Installation</u>. All Customer-owned electrical installations or changes should be inspected upon completion by a competent inspecting authority to insure that wiring, grounding, fixtures and devices have been installed in accordance with the National Electrical Code and such local rules as may be in effect. Where governmental inspection is required by local rules or ordinances, the Company cannot render service until such inspection has been made and formal notice of approval has been received by the Company from the inspecting authority. Where governmental inspection is not required, and before service is rendered by the Company, the Customer shall certify to the Company in writing, that such electrical installation has been inspected by a licensed electrician and is in compliance with all applicable rules and codes in effect. Thereafter, acceptance and receipt of service by the Customer shall constitute certification that the Customer has met all inspection requirements, complied with all applicable codes and rules and, subject to section 2.7 Indemnity to Company, or section 2.71 Indemnity to Company – Governmental, FPL's General Rules and Regulations, the Customer releases, holds harmless and agrees to indemnify the Company from and against loss or liability in connection with the provision of electrical services to or through such Customer-owned electrical installations. The Company reserves the right to inspect the Customer's installation prior to rendering service and from time to time thereafter, but assumes no responsibility whatsoever for any portion thereof.

<u>4.5 Electric Generators</u>. Improper connection of a Customer's generator (or other source of electric service) with the Company's facilities may energize the Company's lines and endanger the lives of the employees, agents or representatives of the Company who may be working on them. Furthermore, such improper connection can seriously damage the Customer's wiring and generator. In order to guard against these dangers, the Company will not connect its service to a Customer's wiring where generators are located unless the wiring conforms to the Company's specifications. These specifications are available on request.

<u>4.6 Momentary Parallel Operation</u>. Permissible and available in all territory served by the Company for electric service to any Customer, at a single point of delivery, when electric service requirements for the Customer's load (i) are supplied or supplemented from the Customer's generation during periods of outages and power ordinarily supplied by the Company, and (ii) necessitate that the Customer's generation operate momentarily in parallel with the Company's system to enable the Customer to transfer its load from the Company's source to the Customer's generation in order to continue the uninterrupted flow of power to the Customer's load. The charge for power supplied by the Company during periods of momentary parallel operation is included in the charge for electric service at the applicable retail rate schedule. No Customer to whom this Rule 4.6 applies shall operate its generation momentarily in parallel with the Company's system unless and until the Customer has entered into a Momentary Parallel Operation Interconnection Agreement with the Company.

5 COMPANY'S INSTALLATIONS

<u>5.1 Protection of Company's Property</u>. The Customer shall properly protect the Company's property on the Customer's premises, and shall permit no one but the Company's agents, or persons authorized by law, to have access to the Company's wiring, meters, and apparatus.

<u>5.2 Damage to Company's Property</u>. In the event of any loss or damage to property of the Company caused by or arising out of carelessness, neglect or misuse by the Customer, the cost of making good such loss or repairing such damage shall be paid by the Customer.

<u>5.3 Relocation of Company's Facilities</u>. When there is a change in the Customer's operation or construction which, in the judgment of the Company, makes the relocation of Company's facilities necessary, or if such relocation is requested by the Customer, the Company will move such facilities at the Customer's expense to a location which is acceptable to the Company.

5.4 Attachments to Poles. The use of the Company's poles, wires, towers, structures or other facilities for the purpose of fastening or supporting any radio or television aerials or other equipment, or any wires, ropes, signs, banners or other things, not necessary to the supplying by the Company of electric service to the community, or the locating of same in such proximity to the Company's property or facilities as to cause, or be likely to cause, interference with the supply of electric service, or a dangerous condition in connection therewith, is prohibited, and the Company shall have the right forthwith to remove same without notice. The violator of these rules is liable for any damage resulting therefrom.

5.5 Interference with Company's Facilities. The Customer should not allow trees, vines and shrubs to interfere with the Company's adjacent overhead conductors, service wires, pad mounted transformers and meter. Such interference may result in an injury to persons, or may cause the Customer's service to be interrupted. In all cases the customer should request the Company to trim or remove trees and other growth near the Company's adjacent overhead wires, and under no circumstances should the Customer undertake this work himself, except around service cables when specifically authorized by and arranged with the Company.

5.6 Unobstructed Access to Company's Facilities. The Company shall have perpetual unobstructed access to its overhead and underground facilities such as poles, underground cables, pad mounted transformers and meters in order to perform repair and maintenance in a safe, timely and cost-efficient manner. The Customer is responsible for contacting the Company for guidance before constructing any items which may obstruct the Company's access. Such items include, but are not limited to, building additions, decks, patios, pools, fences or pavings. Relocation of the Company's facilities requiring repair or maintenance, the Company will explore with the Customer all alternatives deemed feasible by the Company to determine the method of repair most acceptable to the Customer. When the most acceptable or only option involves the Customer removing the obstruction or the Customer taking other actions, the Customer shall accomplish the work within 20 working days. Should the Customer fail to accomplish said work within 20 working days or to make other satisfactory arrangements with the Company, the Company may elect to discontinue service to the Customer, pursuant to F.A.C. Rule 25-6.105 (5) (f). In all cases, the Customer will be responsible for all costs in excess of a standard, unobstructed repair.

6 SECURITY DEPOSITS/GUARANTIES

6.1 Security Deposit/Guaranty.

- (1) Before the Company renders service, or upon termination of an existing Unconditional Guaranty Contract, or a surety bond or an irrevocable bank letter of credit, each applicant will be required to provide:
 - a) a Security Deposit consisting of cash, surety bond, or irrevocable bank letter of credit; or
 - b) a guaranty satisfactory to the Company to secure payment of bills; or
 - c) information which satisfies the Company's application requirements for no deposit.
- (2) a) New service Requests If a Security Deposit is required, the Security Deposit for a new service request shall be based upon no more than two months of projected charges, calculated by adding the 12 months of projected charges, dividing this total by 12, and multiplying the result by 2. After the new account has had continuous service for a twelve (12) month period, the amount of the required deposit shall be recalculated using actual data. If an excess deposit is identified by this recalculated amount indicates a deficiency in the deposit held, the utility may bill customer for the difference. Each applicant that provides a guaranty, surety bond, or an irrevocable bank letter of credit as a Security Deposit must enter into the agreement(s) set forth in Tariff Sheet No. 9.400 /9.401 or 9.410 /9.411/9.412 for the guaranty contract, No. 9.440/ 9.441 for the surety bond and 9.430/9.431 and 9.435 for the bank letter of credit.

(Continue on Sheet No. 6.050)

b) Existing Accounts - For an existing account, the total deposit may not exceed 2 months of average actual charges calculated by adding the monthly charges from the 12-month period immediately before the date any change in the deposit amount is sought, dividing this total by 12, and multiplying the result by 2. If the account has less than 12 months of actual charges, the deposit shall be calculated by adding the available monthly charges, dividing this total by the number of months available, and multiplying the result by 2.

<u>6.2 Deposit Interest.</u> The interest due will be paid once a year, ordinarily as a credit on regular bills, and on final bills when service is discontinued. No interest will be paid if service is ordered disconnected for any cause within six months from the date of initial service.

6.21 Residential Deposits. Simple interest at the rate of 2% per annum will be paid to residential Customers for cash deposits when held by the Company.

<u>6.22 Nonresidential Deposits.</u> Simple interest at the rate of 2% per annum will be paid on cash deposits of nonresidential customers. However, simple interest at the rate of 3% per annum will be paid on cash deposits of nonresidential Customers provided the Customer has had continuous service for a period of not less than 23 months, and has not in the preceding 12 months: a) made more than one late payment of the bill (after the expiration of 20 days from the date of mailing or delivery by the Company), b) paid with a check refused by a bank, c) been disconnected for nonpayment at any time, d) tampered with the electric meter, or e) used service in a fraudulent or unauthorized manner.

<u>6.3 Refund of Cash Deposit/Release of Other Security or Guaranty.</u> After a residential Customer has established a prompt payment record and has had continuous service for a period of not less than 23 months, the Company will no longer require a Security Deposit or guaranty for that account, provided the Customer has not, in the preceding twelve (12) months: a) made more than one (1) late payment of the bill (after the expiration of 20 days from the date of mailing or delivery by the Company), b) paid with a check refused by a bank, c) been disconnected for non-payment, or, at any time d) tampered with the electric meter, or e) used service in a fraudulent or unauthorized manner. When the Company no longer requires a Security Deposit or guaranty because the residential Customer meets these terms or because the Customer closes the service account and the Company has received final payment for all bills for service incurred at the account, any cash deposit held by the Company for that account will be refunded, and the obligors on any surety bond, irrevocable letter of credit or guaranty for that account will be refunded for the customer whose name appears thereon. Refunds of cash deposits may be conditioned by the Company upon a showing of proper identification by the person seeking the refund that the individual is the Customer whose name appears on the service account. The utility may elect to refund nonresidential deposits.

<u>6.4 Transfer of Security Deposit/Guaranty</u>. A Customer moving from one service address to another may have the Security Deposit transferred from the former to the new address. If the Security Deposit at the former service address is more or less than required by Rule 6.1 for the new address, the amount of the Security Deposit may be adjusted accordingly. Guaranties may not be transferred to a new service address; however, the guarantor may enter into a new guaranty contract (Tariff Sheet No. 9.400 or 9.410) for the new service address.

7 BILLING

7.1 Billing Periods.

7.11 Regular Bills. Regular bills for service will be rendered monthly. Bills are due when rendered and shall be considered as received by the Customer when delivered or mailed to the service address or some other place mutually agreed upon.

<u>7.12 Prorated Bill</u>. If the billing period is less than 25 days or more than 35 days, the bill will be prorated pursuant to F.S. 366.05(1) (b). A billing period that exceeds 35 days will be calculated as a separate standard billing period as referenced in section 7.13 of FPL's General Rule and Regulations Tariff. A separate bill calculation for the remaining kWh consumption will begin with the application of the lower tiered rate. Should service be disconnected within less than a month from date of connection, the amount billed will not be less than the regular monthly minimum bill.

 $\frac{7.13 \text{ Month.}}{30 \text{ days, more or less.}}$ As used in these Rules and Regulations, a month is an interval between successive regular meter reading dates, which interval may be

7.14 Budget Billing.

<u>7.14.1 Residential.</u> Any residential Customer who has no delinquent balances with the Company is eligible to participate in the Budget Billing Plan described below for RS-1 rate billings. A Customer may terminate participation in the Budget Billing Plan at any time and may be terminated from the Budget Billing Plan by FPL if the Customer becomes subject to collection action on this service account. Once a Customer's participation in the Budget Billing Plan has terminated he/she may not rejoin the Budget Billing Plan for twelve (12) months following the date of termination. Each eligible Customer not on this Budget Billing Plan will be notified annually of its availability.

Under the Budget Billing Plan, a Customer is billed monthly on a levelized consumption basis rather than on the basis of current consumption. The levelized amount is determined by averaging the last 12 monthly billings for the premise, or the average of all available billing history, whichever is less, and applying the current RS-1 rate and appropriate adjustments. If the Customer has not resided at the premise for 12 months, the Customer's monthly billings plus the previous tenant's billings will be used. Any difference between the levelized amount and the regular bill amount is added to a deferred balance. The current levelized amount is adjusted each month by adding the deferred balance adjustment, which is calculated by dividing the current deferred balance total by 12. The levelized amount, plus the deferred balance adjustment, constitutes the current month's Budget Billing amount. Customers on the Budget Billing Plan will receive the following information on their monthly bill: current consumption and associated charges, the total budget bill charge, and the cumulative deferred balance.

If the Customer's participation in the Budget Billing Plan is terminated, any amount in the deferred balance which the Customer owes to FPL will be billed to the Customer according to the terms of Section 7.9; any amount in the deferred balance which is owed to the Customer will be credited against any outstanding billed amounts, and any remaining balance will be credited against the Customer's future billings or returned upon request. Customers who transfer the location of their service account within FPL's service territory will have the debit or credit balance transferred to the new service address.

7.14.2 Non-residential. Any GS-1 or GSD-1 Customer who has no delinquent balances and has been at the same location for 12 consecutive months with the Company is eligible to participate in the Budget Billing Plan described below for GS-1 and GSD-1 rate billings. However, GS-1 or GSD-1 Customers that rent electrical facilities from the Company under a Facility Rental Service Agreement will not be eligible to participate in this Budget Billing Plan. Additionally, GSD-1 customers taking service under the Seasonal Demand Time of Use Rider will not be eligible to participate in the Budget Billing Plan. A Customer may terminate participation in the Budget Billing Plan at any time and may be terminated from the Budget Billing Plan by FPL if the Customer becomes subject to collection action on this service account. Once a Customer's participation in the Budget Billing Plan has terminated he/she may not rejoin the Budget Billing Plan for twelve (12) months following the date of termination. Each eligible Customer not on this Budget Billing Plan will be notified annually of its availability.

Under the Budget Bill Plan, a Customer is billed monthly on a levelized consumption basis rather than on the basis of current consumption. The levelized amount is determined by averaging the last 12 monthly billings for the premise and applying the current GS-1 or GSD-1 rate and appropriate adjustments. If the Customer has not received electric service at the premise for 12 consecutive months, the Customer is not eligible to participate in the program. Any difference between the levelized amount and the regular bill amount is added to a deferred balance. The current levelized amount is adjusted each month by adding the deferred balance adjustment, which is calculated by dividing the current deferred balance total by 12. The levelized amount, plus the deferred balance adjustment, constitutes the current month's Budget Billing amount. Customers on the Budget Bill Plan will receive the following information on their monthly bill: current consumption and associated charges, the total budget bill charge, and the cumulative deferred balance.

If the Customer's participation in the Budget Bill Plan is terminated either at the request of the Customer or the Company, or as a result of termination of this Budget Billing Plan, any amount in the deferred balance which the Customer owes to FPL will be billed to the Customer according to the terms of Section 7.9; any amount in the deferred balance which is owed to the Customer will be credited against any outstanding billed amounts and any remaining balance will be credited against the Customer's future billings or returned upon request. Customers who transfer the location of their service account within FPL's service territory will have the debit or credit balance transferred to the new service address.

7.15 Summary Billing. A customer with ten (10) or more FPL accounts may request a single statement for the billing and payment of those accounts under Summary Billing. With Summary Billing, the Customer designates the accounts to be included and the cycle day each month when the Summary Bill is to be rendered. FPL will read each meter and calculate the billing amount for each account separately. The billing amount for each of the designated accounts will be totaled on the Summary Billing Statement, with each of the individual account bills attached as backup. Summary Bills are due when rendered and Customers are subject to removal from the program if bills are not paid within ten (10) days from the date of mailing.

<u>7.2 Non-Receipt of Bills.</u> Non-receipt of bills by the Customer shall not release or diminish the obligation of the Customer with respect to payment thereof.

<u>7.3 Evidence of Consumption</u>. When service used is measured by meters, the Company's accounts thereof shall be accepted and received at all times, places and courts as prima facie evidence of the quantity of electricity used by the Customer unless it is established that the meter is not accurate within the limits specified by the Commission.

<u>7.4 Application of Rate Schedules</u>. Electric service will be measured by a single metering installation for each point of delivery. The Company will establish one point of delivery for each Customer and calculate the bill accordingly. Two or more points of delivery shall be considered as separate services and bills separately calculated for each point of delivery.

The Company may adjust the measured kilowatt-demand (kwd) of a Customer to compensate for registration of an abnormal demand level due to testing of electrically-operated equipment prior to general operation provided that the Customer contacts the Company in advance and schedules the testing at a mutually agreed upon time.

7.5 Optional Rate. Where a Customer is eligible to take service at a given location under one of two or more optional rate schedules, the Company will, on request, assist in the selection of the most advantageous rate on an annual basis. If the Customer applies in writing for another applicable schedule, the Company will bill on such elected schedule from and after the date of the next meter reading. However, a Customer having made such a change of rate may not make another change until an interval of twelve (12) months has elapsed.

<u>7.6 Taxes and Charges</u>. All of the Company's rates, including minimum and demand charges and service guarantees, are dependent upon Federal, State, County, Municipal, District, and other Governmental taxes, license fees or other impositions, and may be increased or a surcharge added if and when the cost per kilowatt hour, or per Customer, or per unit of demand or other applicable unit of charge, is increased because of an increase in any or all such taxes, license fees or other impositions. A franchise charge shall be added to the bills of all Florida Public Service Commission jurisdictional customers, as determined by the franchise agreements between Florida Power & Light Company and governmental authorities. The charge shall be computed as a percentage of the bill for energy including fuel delivered within the franchise area, excluding separately stated taxes and the franchise charge itself. This charge shall reflect the estimated amount of the annual franchise payment to that specified governmental authority in which the Customer's account is located, plus adjustment for the gross receipts tax and the regulatory assessment fee, and shall be corrected at least annually for any differences between the actual collections and actual payments.

7.7 Disconnection and Reconnection of Service.

<u>7.71 Disconnection of Service</u>. When discontinuing electric service, Customers should notify the Company at least one (1) business day prior to the requested discontinuation date. Customers are responsible for all electric service used on the premise until notice is received and the Company has had a reasonable time to discontinue service. A billing address should be provided to the Company for issuance of the final billing statement and/or deposit refund. When a Customer orders service discontinued, the Company may ask the Customer to open the main switch upon vacating the premises. This will allow the use of electric service until the time of departure and will insure that no energy is used or charges accrue after the Customer leaves. As convenient, a Company employee will visit the premises to read the meter.

7.72 Reconnection of Service. A Customer who reconnects service by closing the switch should give immediate notice thereof to the Company so that proper records may be maintained. Should the Customer neglect to give such notice, the Company's representative will note the reconnection and it will be recorded as of the date when the switch was closed. If this date cannot be readily determined, reconnection shall be recorded as of the preceding meter reading date.

<u>7.8 Change of Occupancy</u>. When a change of occupancy takes place on any premises supplied by the Company with electric service, notice shall be given to the Company not less than one (1) business day prior to the date of change. The outgoing party will be held responsible for all electric service used on such premises until such notice is received and the Company has had a reasonable time to discontinue service. However, if such notice has not been received by the Company prior to the date of change, the accepted application of the succeeding occupant for the electric service will automatically terminate the prior account.

<u>7.9 Delinquent Bills</u>. Bills are due when rendered and become delinquent if not paid within twenty (20) days from the mailing or delivery date. Thereafter, following five (5) working days' written notice, service may be discontinued and the deposit applied toward settlement of the bill. For purposes of this subsection, "working day" means any day on which the Company's business offices are open and the U.S. Mail is delivered.

8 METERS

<u>8.1 Location of Meters.</u> The Company will determine the location of and install and properly maintain at its own expense such standard meter or meters and metering equipment as may be necessary to measure the electric service used by the Customer. The Customer will keep the meter location clear of obstructions at all times in order that the meter may be read and the metering equipment may be maintained or replaced. If a Customer requests a different location for meter placement from that designated by the Company on initial application for service and the Company agrees that the different meter location is acceptable to the Company, the Customer shall pay the incremental cost of installing the meter at the different meter location. If an existing Customer requests relocation of an existing installed meter and the Company agrees that the different meter the different meter location is acceptable to the Company, the existing Customer shall pay the incremental cost of relocating the meter at the different meter at the different meter location is acceptable to the Company, the existing Customer shall pay the incremental cost of relocating the meter at the different meter location.

<u>8.2 Setting and Removing Meters.</u> None but duly authorized agents of the Company or persons authorized by law shall set or remove, turn on or turn off, or make any changes which will affect the accuracy of such meters. Connections to the Company's system are to be made only by its employees.

<u>8.3 Tampering with Meters.</u> Title to meters and metering equipment shall be and remain in the Company. Unauthorized connections to, or tampering with the Company's meter or meters, or meter seals, or indications or evidence thereof, subjects the Customer to immediate discontinuance of service, prosecution under the laws of Florida, adjustment of prior bills for services rendered, a tampering penalty of \$200 for residential and non-demand commercial customers and \$1,000 for all other customers, and reimbursement to the Company for all extra expenses incurred on this account.

<u>8.4 Meter Tests</u>. The Company employs every practicable means to maintain the commercial accuracy of its meters. Meter tests, and billing adjustments for inaccurate meters, are in accordance with the methods and procedure prescribed by the Florida Public Service Commission.

<u>8.5 Failure of Meter</u>. When a meter fails, or part or all of the metering equipment is destroyed, billing will be estimated based upon the registration of check metering equipment or other available data.

9 SERVICE STANDARDS

These "General Rules and Regulations for Electric Service" include, by reference, the terms and provisions of the Company's currently effective "Electric Service Standards" on file with the Florida Public Service Commission and is available on request. The "Standards" are primarily concerned with the electrical facilities and related equipment prior to installation and use. They explain the general character of electric service supplied, the meters, and other devices furnished by the Company, and the wiring and apparatus provided and installed by the Customer. The Standards serve as a guide to architects, engineers, electrical dealers and contractors in planning, installing, repairing or renewing electrical installations.

INSTALLATION OF UNDERGROUND ELECTRIC DISTRIBUTION FACILITIES TO SERVE RESIDENTIAL CUSTOMERS

SECTION 10.1 DEFINITIONS

The following words and terms, when used in Section 10, shall have the meaning indicated:

<u>APPLICANT</u> - Any person, partnership, association, corporation, or governmental agency controlling or responsible for the development of a new subdivision or dwelling unit who applies for the underground installation of distribution facilities.

BACKBONE - The distribution system excluding feeder and that portion of the service lateral which is on the lot being served by that service lateral.

<u>BUILDING</u> - Any structure designed for residential occupancy, excluding a townhouse unit, which contains less than five individual dwelling units.

<u>CABLE IN CONDUIT SYSTEM</u> - Underground residential distribution systems where all underground primary, secondary, service and street light conductors are installed in direct buried conduit. Other facilities associated with cable in conduit, such as transformers, may be above ground.

<u>COMMISSION</u> - The Florida Public Service Commission.

COMPANY - The Florida Power & Light Company.

<u>DISTRIBUTION SYSTEM</u> - Electric service facilities consisting of primary and secondary conductors, service laterals, conduits, transformers, and necessary accessories and appurtenances for the furnishing of electric power at utilization voltage.

<u>DWELLING UNIT</u> – A single unit providing complete, independent living facilities for one or more persons including permanent provisions for living, sleeping, eating, cooking, and sanitation.

FEEDER MAIN - A three-phase primary installation, including switches, which serves as a source for primary laterals and loops through suitable overcurrent devices.

FINAL GRADE - The ultimate elevation of the ground, paved or unpaved, which will prevail in a subdivision or tract of land.

<u>MOBILE HOME (TRAILER)</u> - A vehicle or conveyance, permanently equipped to travel upon the public highways, that is used either temporarily or permanently as a residence or living quarters.

<u>MULTIPLE-OCCUPANCY BUILDING</u> - A structure erected and framed of component structural parts and designed to contain five or more individual dwelling units.

<u>OVERHEAD SYSTEM</u> - Distribution system consisting of primary, secondary and service conductors and aerial transformers supported by poles.

POINT OF DELIVERY - The point where the Company's wires or apparatus are connected to those of the Customer. See Section 10.2.11.

<u>PRIMARY LATERAL</u> - That part of the electric distribution system whose function is to conduct electricity at the primary level from the feeder main to the transformers. It usually consists of a single-phase conductor or insulated cable, with conduit, together with necessary accessory equipment for supporting, terminating and disconnecting from the primary mains by a fusable element.

<u>SERVICE LATERAL</u> - The entire length of underground service conductors and conduit between the distribution source, including any risers at a pole or other structure or from transformers, from which only one point of service will result, and the first point of connection to the Service Entrance Conductors in a terminal or meter box outside the building wall.

<u>SERVICE ENTRANCE CONDUCTORS</u> – The Customer's conductors from point of connection at the service drop or service lateral to the service equipment.

(Continued on Sheet No. 6.085)

(Continued from Sheet No. 6.080)

<u>SUBDIVISION</u> - The tract of land which is divided into five or more building lots or upon which five or more separate dwelling units are to be located, or the land on which is to be constructed new multiple-occupancy buildings.

<u>TOWNHOUSE</u> - A one-family dwelling unit of a group such that units are separated only by fire walls. Each townhouse unit shall be constructed upon a separate lot and serviced with separate utilities and shall otherwise be independent of one another.

<u>TUG</u> - An acronym formed from the term Temporary UnderGround used to describe the temporary condition in which a building's permanent underground FPL service lateral is utilized to provide electric service to that building during its construction.

SECTION 10.2 GENERAL

10.2.1. Application

Underground electric distribution facilities are offered in lieu of overhead facilities in accordance with these Rules and Regulations for:

- a) New Residential Subdivisions and Developments.
- b) New Service Laterals from Overhead Systems.
- c) Replacement of Existing Overhead and Underground Service Laterals.
- d) New Multiple-Occupancy Residential Buildings.

10.2.2. Early Notification and Coordination

In order for the Company to provide service when required, it is necessary that the Applicant notify the Company during the early stages of planning major projects. Close coordination is necessary throughout the planning and construction stages by the Company, the architect, the builder, the subcontractors and the consulting engineer to avoid delays and additional expense. Particular attention must be given to the scheduling of the construction of paved areas and the various subgrade installations of the several utilities. Failure of the Applicant to provide such notification and coordination shall result in the Applicant paying any additional costs incurred by the Company.

10.2.3. Changes to Plans, Layout or Grade

The Applicant shall pay for any additional costs imposed on the Company by Applicant including, but not limited to, engineering design, administration and relocation expenses, due to changes made subsequent to the agreement in the subdivision or development layout or final grade.

10.2.4. Underground Installations Not Covered

Where the Applicant requests or governmental ordinance mandates underground electric facilities including -but not limited to - three phase primary feeder mains, transformers, pedestal mounted terminals, switching equipment, meter cabinets, service laterals or other electric facilities not specifically covered by these Rules and Regulations and where overhead facilities would otherwise be provided, the Applicant shall pay the Company the differential installed cost between the underground facilities and the equivalent overhead facilities as calculated by the Company. The Applicant shall also provide necessary rights of way and easements as given in Section 10.2.7.

10.2.5. Type of System Provided

The costs quoted in these rules are for underground residential distribution service laterals, secondary and primary conductors of standard Company design with cable in conduits and above-grade appurtenances. Unless otherwise stated, service provided will be 120/240 volt, single phase. If other types of facilities other than standard Company design are requested by the Applicant or required by governmental authority, the Applicant will pay the additional costs, as calculated by the Company, if any.

10.2.6. Design and Ownership

The Company will design, install, own, and maintain the electric distribution facilities up to the designated point of delivery except as otherwise noted. Any payment made by the Applicant under the provisions of these Rules will not convey to the Applicant any rights of ownership or right to specify Company facilities utilized to provide service.

10.2.7. Rights of Way and Easements

The Applicant shall record and furnish satisfactory rights of way and easements, including legal descriptions of such easements and all survey work associated with producing legal descriptions of such easements, as required by and at no cost to the Company prior to the Company initiating construction. Before the Company will start construction, these rights of way and easements must be cleared by the Applicant of trees, tree stumps and other obstructions that conflict with construction, staked to show property corners and survey control points, graded to within six inches of final grade, with soil stabilized. In addition, the Applicant shall provide stakes showing final grade along the easement. Such clearing and grading must be maintained by the Applicant during construction by the utility.

10.2.8. Contributions and Credits

The Applicant shall pay the required contribution upon receipt of written notification from the Company. No utility construction shall commence prior to execution of the Underground Distribution Facilities Installation Agreement set forth in Tariff Sheet Nos. 9.700, 9.701 and 9.702 and payment in full of the entire contribution. Where, by mutual agreement, the Applicant performs any of the work normally performed by the Company, the Applicant shall receive a credit for such work in accordance with the credit amounts contained herein, provided that the work is in accordance with Company specifications. Such credit shall not exceed the total differential costs. The credit will be granted after the work has been inspected by the Company and, in the case of Applicant-installed conduit, after the applicable conductors have been installed.

(Continued on Sheet No. 6.095)

(Continued from Sheet No. 6.090)

10.2.8.1 Credit for TUGs

If the Applicant installs the permanent electric service entrance such that FPL's service lateral can be subsequently installed and utilized to provide that building's construction service, the Applicant shall receive a credit in the amount of \$70.12 per service lateral, subject to the following requirements:

- a) TUGs must be inspected and approved by the local inspecting authority.
- b) All service laterals within the subdivision must be installed as TUGs.
- c) FPL must be able to install the service lateral, energize the service lateral, and set the meter to energize the load side of the meter can, all in a single trip. Subsequent visits other than routine maintenance or meter readings will void the credit.
- d) Thereafter, acceptance and receipt of service by the Customer shall constitute certification that the Customer has met all inspection requirements, complied with all applicable codes and rules and, subject to section 2.7 Indemnity to Company, or section 2.71 Indemnity to Company – Governmental, FPL's General Rules and Regulations, the Customer releases, holds harmless and agrees to indemnify the Company from and against loss or liability in connection with the provision of electrical services to or through such Customer-owned electrical installations.
- e) The Applicant shall be held responsible for all electric service used until the account is established in the succeeding occupant's name.

This credit applies only when FPL installs the service - it does not apply when the applicant installs the service conduits, or the service conduits and cable.

10.2.9. Location of Distribution Facilities

Underground distribution facilities will be located, as determined by the Company, to maximize their accessibility for maintenance and operation. The Applicant shall provide accessible locations for meters when the design of a dwelling unit or its appurtenances limits perpetual accessibility for reading, testing, or making necessary repairs and adjustments.

10.2.10. Special Conditions

The costs quoted in these rules are based on conditions which permit employment of rapid construction techniques. The Applicant shall be responsible for necessary additional hand digging expenses other than what is normally provided by the Company. The Applicant is responsible for clearing, compacting, boulder and large rock removal, stump removal, paving, and addressing other special conditions. Should paving, grass, landscaping or sprinkler systems be installed prior to the construction of the underground distribution facilities, the Applicant shall pay the added costs of trenching and backfilling and be responsible for restoration of property damaged to accommodate the installation of underground facilities.

10.2.11. Point of Delivery

The point of delivery shall be determined by the Company and will normally be at or near the part of the building nearest the point at which the secondary electric supply is available to the property. When a location for a point of delivery different from that designated by the Company is requested by the Applicant, and approved by the Company, the Applicant shall pay the estimated full cost of service lateral length, including labor and materials, required in excess of that which would have been needed to reach the Company's designated point of service. The additional cost per trench foot is \$7.91. Where an existing trench is utilized, the additional cost per trench foot is \$3.00. Where the Applicant provides the trenching, installs Company provided conduit according to Company specifications and backfilling, the cost per additional trench foot is \$2.16. Any re-designation requested by the Applicant shall conform to good safety and construction practices as determined by the Company. Service laterals shall be installed, where possible, in a direct line to the point of delivery.

(Continued on Sheet No. 6.096)

(Continued from Sheet No. 6.095)

10.2.12. Location of Meter and Downpipe

The Applicant shall install a meter enclosure and downpipe to accommodate the Company's service lateral conductors at the point designated by the Company. These facilities will be installed in accordance with the Company's specifications and all applicable codes.

10.2.13. <u>Relocation or Removal of Existing Facilities</u>

If the Company is required to relocate or remove existing facilities in the implementation of these Rules, all costs thereof shall be borne exclusively by the Applicant, as follows:

- a) For removal of existing facilities, these costs will include the costs of removal, the in-place value (less salvage) of the facilities so removed and any additional costs due to existing landscaping, pavement or unusual conditions.
- b) For relocation of existing facilities, these costs will include the costs of relocation of reusable equipment, costs of removal of equipment that cannot be reused, costs of installation of new equipment, and any additional costs due to existing landscaping, pavement or unusual conditions.

10.2.14. Development of Subdivisions

The Tariff charges are based on reasonably full use of the land being developed. Where the Company is required to construct underground electric facilities through a section or sections of the subdivision or development where full use of facilities as determined by the Company, will not be experienced for at least two years, the Company may require a deposit from the Applicant before construction is commenced. This deposit, to guarantee performance, will be based on the estimated total cost of such facilities rather than the differential cost. The amount of the deposit, without interest, less any required contributions will be returned to the Applicant on a prorata basis at quarterly intervals on the basis of installations to new customers. Any portion of such deposit remaining unrefunded, after five years from the date the Company is first ready to render service from the extension, will be retained by the Company.

10.2.15 <u>Service Lateral Conductor</u>

All residential Tariff charges are based on a single service conductor installed in a single 2" conduit, limited to a maximum size of 4/0 aluminum. All parallel services, or any single services requiring service conductor larger than 4/0 aluminum, require additional charges determined by specific cost estimate.

FLORIDA POWER & LIGHT COMPANY

SECTION 10.3 UNDERGROUND DISTRIBUTION FACILITIES FOR RESIDENTIAL SUBDIVISIONS AND DEVELOPMENTS

10.3.1. Availability

When requested by the Applicant, the Company will provide underground electric distribution facilities, other than for multiple occupancy buildings, in accordance with its standard practices in:

- a) Recognized new residential subdivision of five or more building lots.
- b) Tracts of land upon which five or more separate dwelling units are to be located.

For residential buildings containing five or more dwelling units, see SECTION 10.6 of these Rules.

10.3.2. Contribution by Applicant

a) The Applicant shall pay the Company the average differential cost for single phase residential underground distribution service based on the number of service laterals required or the number of dwelling units, as follows:

1.	Where density is 6.0 or more dwelling units per acre:		plicant's <u>tribution</u>
	 1.1 Buildings that do not exceed four units, townhouses, and mobile homes – per service lateral. 1. Subdivisions with 300 or more total service laterals 2. Subdivisions from 100 to 299 total service laterals 3. Subdivisions less than 100 total service laterals 	\$ \$ \$	0.00 0.00 0.00
	 1.2 Mobile homes having Customer-owned services from meter center installed adjacent to the FPL primary trench route - per dwelling unit. 1. Subdivisions with 300 or more total dwelling units 2. Subdivisions from 100 to 299 total dwelling units 3. Subdivisions less than 100 total dwelling units 	\$ \$ \$	0.00 0.00 0.00
2.	Where density is 0.5 or greater, but less than 6.0 dwelling units per acre:		
	 Buildings that do not exceed four units, townhouses, and mobile homes – per service lateral 1. Subdivisions with 200 or more total service laterals 2. Subdivisions from 85 to 199 total service laterals 3. Subdivisions less than 85 total service laterals 	\$ \$ \$	$0.00 \\ 0.00 \\ 0.00$

3. Where the density is less than 0.5 dwelling units per acre, or the Distribution System is of non-standard design, individual cost estimates will be used to determine the differential cost as specified in Paragraph 10.2.5.

Additional charges specified in Paragraphs 10.2.10 and 10.2.11 may also apply.

b) The above costs are based upon arrangements that will permit serving the local underground distribution system within the subdivision from overhead feeder mains. If feeder mains within the subdivision are deemed necessary by the Company to provide and/or maintain adequate service and are required by the Applicant or a governmental agency to be installed underground, the Applicant shall pay the Company the average differential cost between such underground feeder mains within the subdivision and equivalent overhead feeder mains, as follows:

	Applicant's <u>Contribution</u>
Cost per foot of feeder trench within the subdivision	
(excluding switches)	\$10.09
Cost per above ground padmounted switch package	\$25,716.84

(Continued on Sheet No. 6.110)

(Continued from Sheet No. 6.100)

c) Where primary laterals are needed to cross open areas such as golf courses, parks, other recreation areas and water retention areas, the Applicant shall pay the average differential costs for these facilities as follows:

Cost per foot of primary lateral trench within the subdivision

1) Single Phase - per foot	\$0.98
2) Two Phase - per foot	\$3.02
3) Three Phase - per foot	\$4.70

d) For requests for service where underground facilities to the lot line are existing and a differential charge was previously paid for these facilities, the cost to install an underground service lateral to the meter is as follows:

Density less than 6.0 dwelling units per acre:	\$398.76
Density 6.0 or greater dwelling units per acre:	\$295.96

10.3.3. Contribution Adjustments

a) Credits will be allowed to the Applicant's contribution in Section 10.3.2. where, by mutual agreement, the Applicant provides all trenching and backfilling for the Company's distribution system, excluding feeder.

				Credit to Applicant's C	ontribution
	1.	Wh	ere density is 6.0 or more dwelling units per acre:	Backbone	Service
		1.1	Buildings that do not exceed four units, townhouses, and mobile homes		
			- per service lateral.	\$174.32	\$183.00
		1.2	8		
			services from meter center		
			installed adjacent to the FPL primary trench route		
			- per dwelling unit.	\$144.16	N/A
				\$11110	1
	2.	Wh	ere density is 0.5 or greater, but less		
		thar	n 6.0 dwelling units per acre:		
			Buildings that do not exceed four units, townhouses, and mobile homes		
			- per service lateral	\$288.73	\$256.20
b)			will be allowed to the Applicant's contribution in Section 10.3.2 nt installs all Company-provided conduit excluding feeder per F		
	1.	Wh	ere density is 6.0 or more dwelling units per acre:	Backbone	Service
		1.1	Buildings that do not exceed four units,	Backbolle	Service
			townhouses, and mobile homes	Ф 70 <i>С</i> 4	Φ <i>Γ</i> <u></u> (00
			- per service lateral.	\$72.54	\$56.09
			(Continued on Sheet No. 6.115)		

(Continued from Sheet No. 6.110)

			Credit to Applicant's Contribution	
			Backbone	Service
	1.2	Mobile homes having Customer-owned services from meter center installed adjacent to the FPL primary trench route - per dwelling unit.	\$59.15	N/A
2.		ere density is .5 or greater, but less than dwelling units per acre, per service lateral.	\$116.25	\$68.71

c) Credits will be allowed to the Applicant's contribution in Section 10.3.2. where, by mutual agreement, the Applicant provides a portion of trenching and backfilling for the Company's facilities, per foot of trench – \$4.07.

d) Credits will be allowed to the Applicant's contribution in section 10.3.2. where, by mutual agreement, the Applicant installs a portion of Company-provided PVC conduit, per FPL instructions (per foot of conduit): 2" PVC - \$0.70; larger than 2" PVC - \$0.98.

e) Credit will be allowed to the Applicant's contribution in section 10.3.2., where, by mutual agreement, the Applicant installs an FPL-provided feeder splice box, per FPL instructions, per box - \$776.87.

f) Credit will be allowed to the Applicant's contribution in section 10.3.2., where by mutual agreement, the Applicant installs an FPL-provided primary splice box, per FPL instructions, per box - \$272.05.

g) Credit will be allowed to the Applicant's contribution in section 10.3.2., where, by mutual agreement, the Applicant installs an FPL-provided secondary handhole, per FPL instructions, per handhole: 17" handhole - \$25.24; 24" or 30" handhole - \$71.52.

h) Credit will be allowed to the Applicant's contribution in section 10.3.2., where, by mutual agreement, the Applicant installs an FPL-provided concrete pad for a pad-mounted transformer or capacitor bank, per FPL instructions, per pad - \$70.12.

i) Credit will be allowed to the Applicant's contribution in Section 10.3.2., where, by mutual agreement, the Applicant installs a portion of Company-provided flexible HDPE conduit, per FPL instructions (per foot of conduit): \$0.14.

j) Credit will be allowed to the Applicant's contribution in Section 10.3.2., where, by mutual agreement, the Applicant installs an FPL-provided concrete pad and cable chamber for a pad-mounted feeder switch, per pad and cable chamber - \$660.48.

SECTION 10.4 UNDERGROUND SERVICE LATERALS FROM OVERHEAD ELECTRIC DISTRIBUTION SYSTEMS

10.4.1. <u>New Underground Service Laterals</u>

When requested by the Applicant, the Company will install underground service laterals from overhead systems to newly constructed residential buildings containing less than five separate dwelling units.

10.4.2. <u>Contribution by Applicant</u>

a) The Applicant shall pay the Company the following differential cost between an overhead service and an underground service lateral, as follows:

		Applicant's Contribution
1.	For any density:	
	Buildings that do not exceed four units, townhouses, and mobile homes	
	a) per service lateral (includes service riser installation)b) per service lateral (from existing handhole or PM TX)	\$756.40 \$398.76
2.	For any density, the Company will provide a riser to a handhole at the base of a pole	\$767.83

Additional charges specified in Paragraphs 10.2.10 and 10.2.11 may also apply. Underground service or secondary extensions beyond the boundaries of the property being served will be subject to additional differential costs as determined by individual cost estimates.

10.4.3. <u>Contribution Adjustments</u>

1.

a) Credit will be allowed to the Applicant's contribution in Section 10.4.2 where, by mutual agreement, the Applicant provides trenching and backfilling for the Company's facilities. This credit is:

	Credit To Applicant's <u>Contribution</u>
For any density:	
Buildings that do not exceed four units, townhouses, and mobile homes - per foot	\$4.07

(Continued on Sheet No. 6.125)

(Continued from Sheet No. 6.120)

b) Credit will be allowed to the Applicant's contribution in Section 10.4.2, where by mutual agreement, the Applicant installs Company-provided conduit, per FPL instructions, as follows:

1. For any density:

Buildings that do not exceed four units, townhouses, and mobile homes - per foot: 2" PVC \$0.70 Larger than 2" PVC \$0.98

c) Credit will be allowed to the Applicant's contribution in Section 10.4.2, where by mutual agreement, the Applicant requests the underground service to be installed as a TUG (subject to the conditions specified in Section 10.2.8.1), per service lateral, as follows:

\$70.12

1. For any density:

Buildings that do not exceed four units, townhouses, and mobile homes -per service lateral:

SECTION 10.5 UNDERGROUND SERVICE LATERALS REPLACING EXISTING RESIDENTIAL OVERHEAD AND UNDERGROUND SERVICES

10.5.1. <u>Applicability</u>

When requested by the Applicant, the Company will install underground service laterals from existing systems as replacements for existing overhead and underground services to existing residential buildings containing less than five individual dwelling units.

10.5.2. Rearrangement of Service Entrance

The Applicant shall be responsible for any necessary rearranging of his existing electric service entrance facilities to accommodate the proposed underground service lateral in accordance with the Company's specifications.

10.5.3 Trenching and Conduit Installation

The Applicant shall also provide, at no cost to the Company, a suitable trench, perform the backfilling and any landscape, pavement or other similar repairs and install Company provided conduit according to Company specifications. When requested by the Applicant and approved by the Company, the Company may supply the trench and conduit and the Applicant shall pay for this work based on a specific cost estimate. Should paving, grass, landscaping or sprinkler systems need repair or replacement during construction, the Applicant shall be responsible for restoring the paving, grass, landscaping or sprinkler systems to the original condition.

10.5.4. Contribution by Applicant

a)		charge per service lateral replacing an existing npany-owned overhead service for any density shall be:	Applicant's
			<u>Contribution</u>
	1.	Where the Company provides an underground service lateral:	\$704.99
	2.	Where the Company provides a riser to a handhole at the base of the pole:	\$1016.79
b)		charge per service lateral replacing an existing Company-owned erground service at Applicant's request for any density shall be:	
	1.	Where the service is from an overhead system:	\$705.62
	2.	Where the service is from an underground system:	\$605.99
c)		charge per service lateral replacing an existing Customer-owned lerground service from an overhead system for any density shall be:	\$456.03
d)	und	charge per service lateral replacing an existing Customer-owned erground service from an underground system for any density l be:	\$98.38

The above charges include conversion of the service lateral from the last FPL pole to the meter location. Removal of any other facilities such as poles, downguys, spans of secondary, etc. will be charged based on specific cost estimates for the requested additional work.

SECTION 10.6 UNDERGROUND SERVICE DISTRIBUTION FACILITIES TO MULTIPLE-OCCUPANCY RESIDENTIAL BUILDINGS

10.6.1. Availability

Underground electric distribution facilities may be installed within the tract of land upon which multiple-occupancy residential buildings will be constructed.

10.6.2. <u>Contribution by Applicant</u>

When feeder mains on tracts of land upon which multiple-occupancy buildings will be constructed are deemed necessary by the Company to provide and/or maintain adequate service, an underground installation is requested by the Applicant, or required by a governmental agency having the authority so to do, the Applicant shall contribute the differential costs provided in Section 10.3.2.b) and 10.3.3.c). There will be no contribution from the Applicant with respect to construction of underground distribution facilities other than feeder mains so long as the Company is free to construct such extensions in the most economical manner, and reasonably full use is made of the tract of land upon which the multiple-occupancy residential buildings will be constructed. Other conditions will require special arrangements.

10.6.3. <u>Responsibility of Applicant</u>

The Applicant shall, at no cost to the Company:

- a) Furnish details and specifications of the proposed building or complex of buildings. The Company will use these in the design of the electric distribution facilities required to render service.
- b) Where the Company determines that transformers are to be located inside the building, the Applicant shall provide in accordance with Company specifications:
 - 1) The vault or vaults necesary for the transformers and associated equipment.
 - 2) The necessary raceways or conduit for the Company's supply cables from the vault or vaults to a suitable point five feet outside the building in accordance with the Company's plans and specifications.
 - Conduits underneath all buildings when required for the Company's supply cables. Such conduits shall extend a minimum of five feet beyond the edge of the buildings for joining to the Company's facilities.
 - 4) The service entrance conductors and raceways from the Applicant's service equipment to the designated point of delivery within the vault.
- c) Where the Company determines that transformers are to be located outside the building, the Applicant shall provide in accordance with Company specifications:
 - 1) The space for padmounted equipment at or near the building, and protective devices for such equipment, if required.
 - 2) The service entrance conductors and raceway from the Applicant's service equipment to the point of delivery designated by the Company at or near the building.
 - 3) Conduits underneath all buildings when required for the Company's supply cables. Such conduits shall extend five feet beyond the edge of the buildings for joining to the Company's facilities.
- d) Provide proper easements, including the right of ingress and egress for the installation, operation and maintenance of the Company's facilities.
- e) Ensure that the metering enclosures are appropriately marked with the same alphabetic or numeric designation used to identify the service address. Such markings shall be of a permanent nature.

10.6.4. Responsibility of the Company

The Company will:

a) Provide the Applicant with the Company's plans to supply the proposed building or complex of buildings, and specifications for the facilities to be provided by the Applicant.

(Continued on Sheet No. 6.150)

(Continued from Sheet No. 6.140)

- b) Furnish and install the primary or secondary conductors from existing or proposed facilities adjoining the property to the point of delivery, together with the ducts, if required, outside the building.
- c) Furnish and install the necessary transformers and associated equipment located either outside the building or in the vault or vaults within the building.
- d) Be solely responsible for the installation, operation and maintenance of all of its facilities.

10.6.5. Service Voltages

The Company will supply service at one of the several secondary voltages available as mutually agreed upon between the Applicant and the Company.

11.0 INSTALLATION OF NEW OR UPGRADED FACILITIES

SECTION 11.1 GENERAL

In accordance with F.A.C. Rule 25-6.064 this tariff section applies to requests for new or upgraded facilities. Nothing herein shall alter the charges or provisions outlined in sections 10 and 13 of this tariff.

An Applicant can be any person, corporation, or entity capable of complying with the requirements of this tariff that has made a request for new or upgraded facilities in accordance with this tariff.

11.1.1 CONTRIBUTION-IN-AID OF CONSTRUCTION (CIAC)

A CIAC shall be required from Applicants requesting new or upgraded facilities prior to construction of the requested facilities based on the formulas presented below.

(a) The CIAC for new or upgraded overhead facilities (CIAC_{OH}) shall be calculated as follows:

CIAC _{OH}	=	Total estimated work
		order job cost of
		installing the facilities

Four years expected incremental base energy revenue

Four years expected incremental base demand revenue, if applicable

- (i) The cost of the service drop and meter shall be excluded from the total estimated work order job cost for new overhead facilities.
- (ii) The net book value and cost of removal, net of the salvage value, for existing facilities shall be included in the total estimated work order job cost for upgrades to those existing facilities.
- (iii)The expected annual base energy and demand charge revenues shall be estimated for a period ending not more than 5 years after the new or upgraded facilities are placed in service.
- (iv) In no instance shall the $CIAC_{OH}$ be less than zero.

(b) The CIAC for new or upgraded underground facilities (CIAC_{UG}) shall be calculated as follows: $CIAC_{UG} = CIAC_{OH} + Estimated difference between the cost of providing the service underground and overhead$

11.1.2 CIAC True-Up

An Applicant may request a one-time review of a paid CIAC amount within 12 months following the inservice date of the new or upgraded facilities. Upon receiving a request, which must be in writing, the Company shall true-up the CIAC to reflect the actual construction costs and a revised estimate of base revenues. The revised estimate of base revenues shall be developed from the actual base revenues received at the time the request is made. If the true-up calculation result is different from the paid CIAC amount, the Company will either issue a refund or an invoice for this difference. This CIAC review is available only to an initial Applicant who paid the original full CIAC amount, not to any other Applicants who may be required to pay a pro-rata share as described in section 11.1.3.

(Continued On Sheet No. 6.200)

(Continued From Sheet No. 6.199)

11.1.3 Proration of CIAC

CIAC is proratable if more Applicants than the Initial Applicant are expected to be served by the new or upgraded facilities ("New Facilities") within the three-year period following the in-service date. The Company shall collect the full CIAC amount from the Initial Applicant. Thereafter, the Company shall collect, and pay to the Initial Applicant, a pro-rata share of the CIAC from each additional Applicant to be served from these New Facilities until the three-year period has expired, or until the number of Applicants served by the New Facilities equals the number originally expected to be served during the three-year period, whichever comes first. Any CIAC or pro-rata share amount due from an Applicant shall be paid prior to construction. For purposes of this tariff, the New Facilities' in-service date is defined as the date on which the New Facilities are installed and service is available to the Initial Applicant, as determined by the Company.

SECTION 11.2 INSTALLATION OF UNDERGROUND ELECTRIC DISTRIBUTION FACILITIES FOR NEW CONSTRUCTION

11.2.0 Distribution System

Electric service facilities consisting of primary and secondary conductors, service drops, service laterals, conduits, transformers and necessary accessories and appurtenances for the furnishing of electric power at utilization voltage.

11.2.1 Application

This tariff section applies to all requests for underground electric distribution facilities where the facilities requested will constitute new construction, other than those requests covered by sections 10, 12 and 13 of this tariff. Any Applicant may submit a request as follows. Requests shall be in writing and must specify in detail the proposed facilities that the Applicant desires to be installed as underground electric distribution facilities in lieu of overhead electric distribution facilities. Upon receipt of a written request FPL will determine the non-refundable deposit amount necessary to secure a binding cost estimate and notify the applicant of said amount. Where system integrity would be compromised by the delay of a system improvement due to the time allowances specified below, said time allowances shall be reduced such that all terms and conditions of this tariff must be met 30 days prior to the date that construction must begin to allow the underground facility to be completed and operable to avert a system compromise.

11.2.2 Contribution-in-Aid-Of-Construction (CIAC)

Upon the payment of a non-refundable deposit by an Applicant, FPL shall prepare a binding cost estimate specifying the contribution-in-aid-of-construction (CIAC) required for the installation of the requested underground distribution facilities in addition to any CIAC required for facilities extension, where the installation of such facilities is feasible, and provide said estimate to the Applicant upon completion of the estimate along with an Underground Distribution Facilities Installation Agreement. The CIAC may be subject to increase or refund if the project scope is enlarged or reduced at the request of the Applicant, or the CIAC is found to have a material error prior to the commencement of construction. The binding cost estimate provided to an Applicant shall be considered expired if the Applicant does not enter into an Underground Distribution Facilities Installation Agreement and pay the CIAC amount specified for the installation of the requested underground electric distribution facilities within 180 days of delivery of the binding cost estimate to the Applicant by FPL.

11.2.3 Non-Refundable Deposits

The non-refundable deposit for a binding cost estimate for a direct buried cable in conduit underground electric distribution system shall be determined by multiplying the number of proposed trench feet for new underground electric distribution facilities to be installed by \$0.75. The deposit must be paid to FPL to initiate the estimating process. The deposit will not be refundable, however, it will be applied in the calculation of the CIAC required for the installation of underground distribution facilities. The deposit and the preparation of a binding cost estimate are a prerequisite to the execution of an Underground Distribution Facilities Installation Agreement. If the request for underground electric distribution facilities involves less than 250 proposed trench feet then no deposit will be required for a binding cost estimate, provided, however, that all other requirements of this tariff shall still apply.

(Continued on Sheet No. 6.210)

(Continued from Sheet No. 6.200)

11.2.4 Non-Binding Cost Estimates

Any person, corporation, or entity may request a non-binding cost estimate free of charge. The non-binding cost estimate shall be an order of magnitude estimate to assist the requestor in determining whether to go forward with a binding cost estimate. An Underground Distribution Facilities Installation Agreement may not be executed on the basis of a non-binding cost estimate.

11.2.5 <u>Underground Distribution Facilities Installation Agreement</u>

Any Applicant seeking the installation of underground distribution facilities pursuant to a written request hereunder shall execute the Underground Distribution Facilities Installation Agreement set forth in this tariff at Sheet Nos. 9.700, 9.701 and 9.702. The Agreement must be executed and the CIAC paid by the Applicant within 180 days of the delivery of the binding cost estimate to the Applicant. Failure to execute the Agreement and pay the CIAC specified in the agreement within the 180 day time limit, or termination of the Agreement, shall result in the expiration of the binding cost estimate. Any subsequent request for underground facilities will require the payment of a new deposit and the presentation of a new binding cost estimate. For good cause FPL may extend the 180 day time limit. Upon execution of the Underground Distribution Facilities Installation Agreement, payment in full of the CIAC specified in the binding cost estimate, and compliance with the requirements of this tariff, FPL shall proceed to install the facilities identified in a timely manner.

11.2.6 Easements

Before the initiation of any project to provide underground electric distribution facilities pursuant to an Underground Distribution Facilities Installation Agreement, the Applicant shall provide to FPL and record, at no cost to FPL, all easements, including legal descriptions of such easements and all survey work associated with producing legal descriptions of such easements and all survey work associated with producing legal descriptions of such easements, specified as necessary by FPL to accommodate the requested underground facilities along with an opinion of title that the easements are valid. Failure to provide the easements in the manner set forth above within 180 days after delivery of the binding cost estimate to the Applicant shall result in the expiration of the binding cost estimate, the return of any CIAC paid, and the termination of any Underground Distribution Facilities Installation Agreement entered into between the Applicant and FPL. Before FPL will commence construction, those rights of way and easements, contained within the boundaries of a development for which the underground electric distribution facilities are to be installed for new service, shall be staked to show property corners and survey control points, graded to within six inches of final grade, with soil stabilized, and also staked to show the final grade along the easement.

11.2.7 Early Notification and Coordination

In order for FPL to provide service when requested, it is necessary that the Applicant notify FPL during the early stages of major project planning. In matters requiring new service extensions close coordination is necessary throughout the planning and construction stages by FPL, the architect, the builder, the subcontractors and the consulting engineer to avoid delays and additional expense. Particular attention must be given to the scheduling of the construction of paved areas and the various subgrade installations of the several utilities. Failure of the Applicant to provide such notification and coordination shall result in the Applicant being responsible for any additional costs incurred by FPL as a result of said failure.

11.2.8 Changes to Plans, Layout or Grade

The Applicant shall pay for any additional costs incurred by FPL due to changes in the development layout or final grade made by the Applicant subsequent to the development layout or final grade information supplied to FPL for the preparation of the binding cost estimate.

11.2.9 Location of Distribution Facilities

Underground distribution facilities will be located, as determined by FPL, to maximize their accessibility for maintenance and operation. Where construction is for the purpose of new service the Applicant shall provide accessible locations for meters when the design of a building or its appurtenances limit perpetual accessibility for reading, testing, or making necessary repairs and adjustments.

11.2.10 Other Terms and Conditions

Through the execution of the Underground Distribution Facilities Installation Agreement found at Tariff Sheet Nos. 9.700, 9.701 and 9.702, the Applicant agrees to the following:

a) The Applicant shall be responsible for all restoration of, repair of, or compensation for, property affected, damaged, or destroyed, to accommodate the installation of underground distribution facilities;

(Continued on Sheet No. 6.220)

(Continued from Sheet No. 6.210)

- (b) subject to section 2.7 Indemnity to Company, or section 2.71 Indemnity to Company Governmental, FPL's General Rules and Regulations, the Applicant shall indemnify FPL from any claim, suit, or other proceeding, which seeks the restoration of, or repair of, or compensation for, property affected, damaged, or destroyed, to accommodate the installation of underground distribution facilities arising from or brought as a result of the installation of underground distribution facilities;
- (c) the Applicant shall clear easements provided to FPL of trees, tree stumps and other obstructions that conflict with construction or installation of underground distribution facilities in a timely manner consistent with FPL's construction schedule.

11.2.11 Type of System Provided

An underground distribution system will be provided in accordance with FPL's current design and construction standards.

11.2.12 Design and Ownership

FPL will design, install, own, and maintain the electric distribution facilities up to the designated point of delivery except as otherwise noted. Any payment made by the Applicant under these Rules will not convey to the Applicant any rights of ownership or right to specify FPL facilities utilized to provide service. The Applicant may, subject to a contractual agreement with FPL, construct and install all or a portion of the underground distribution facilities provide that:

- a) such work meets FPL's construction standards;
- b) FPL will own and maintain the completed distribution facilities;
- c) the construction and installation of underground distribution facilities by the Applicant is not expected to cause the general body of ratepayers to incur greater costs;
- d) the Applicant agrees to pay FPL's current applicable hourly rate for engineering personnel for all time spent reviewing and inspecting the Applicants work done; and
- e) the Applicant agrees to rectify any deficiencies found by FPL prior to the connection of any customers to the underground electric distribution system or the connection of the underground electric distribution facilities to FPL's distribution system. Furthermore, the deficiencies must be corrected in a timely manner or FPL shall perform the construction using overhead facilities and the Applicant will be responsible for paying the cost of installing the overhead facilities and the cost of their removal before the corrected underground facilities will be connected.

INSTALLATION OF UNDERGROUND ELECTRIC DISTRIBUTION FACILITIES FOR THE CONVERSION OF OVERHEAD ELECTRIC DISTRIBUTION FACILITIES

SECTION 12.1 DEFINITIONS

<u>APPLICANT</u> - Any person, corporation, or entity capable of complying with the requirements of this tariff that has made a written request for underground electric distribution facilities in accordance with this tariff.

<u>CONVERSION</u> - Any installation of underground electric distribution facilities where the underground facilities will be substituted for existing overhead electric distribution facilities, including relocations.

<u>CONTRIBUTION-IN-AID-OF-CONSTRUCTION (CIAC)</u> – The CIAC to be paid by an Applicant under this tariff section shall be the result of the following formula:

CIAC =

- 1) The estimated cost to install the requested underground facilities;
- ⁺ 2) The estimated cost to remove the existing overhead facilities; ^a
- + 3) The net book value of the existing overhead facilities; ^a
- 4) The estimated cost that would be incurred to install new overhead facilities, in lieu of underground, to replace the existing overhead facilities (the "Hypothetical Overhead Facilities");
- 5) The estimated salvage value of the existing overhead facilities to be removed; ^a
- + 6) The 30-year net present value of the estimated non-storm underground v. overhead operational costs differential,
- 7) The 30-year net present value of the estimated average Avoided Storm Restoration Costs ("ASRC") calculated as a percentage of the sum of lines 1) through 6). Simplified eligibility criteria for each ASRC Tier are summarized below. Applicants must enter into an Underground Facilities Conversion Agreement with the Company which provides full details on terms, conditions and compliance requirements.

Tier	Percentage	Pole-Line Miles	Customer Conversions	Completion
1 *	25%	3 or more	100%	3 phases
2	10%	1 to <3	100%	3 phases
3	5%	< 1	n/a	n/a

* The GAF Waiver will apply in lieu of Tier 1 ASRC for eligible conversions by Local Government Applicants.

^a In calculating the Applicant's CIAC, elements 2, 3, and 5 of the CIAC formula above are to be excluded from CIAC due from an applicant who submits an application providing a binding notification that said applicant intends to convert existing non-hardened overhead feeder facilities to underground feeder facilities.

GAF Waiver

For Applicants entering into an Underground Facilities Conversion Agreement – Governmental Adjustment Factor Waiver with the Company, the otherwise applicable CIAC amount, as calculated above, shall be reduced by the GAF Waiver. The amount of the GAF Waiver shall be calculated as follows:

GAF Waiver =

25% x the otherwise applicable CIAC;

+ 75% x the ASRC (avoids double-counting the ASRC embedded in the otherwise applicable CIAC.)

If the Applicant elects to construct and install all or part of the underground facilities, then for purposes of calculating the ASRC or the GAF Waiver amount only, the otherwise applicable CIAC shall be adjusted to add FPL's estimated cost for the Applicantperformed work. In addition, the Direct Engineering, Supervision, and Support (DESS) costs associated with this Applicantperformed work will be reduced by 20% from the amount that would have applied if FPL performed this work.

<u>DISTRIBUTION SYSTEM</u> - Electric service facilities consisting of primary and secondary conductors, service drops, service laterals, conduits, transformers and necessary accessories and appurtenances for the furnishing of electric power at utilization voltage.

<u>SERVICE FACILITIES</u> - The entire length of conductors between the distribution source, including any conduit and or risers at a pole or other structure or from transformers, from which only one point of service will result, and the first point of connection to the service entrance conductors at a weatherhead, in a terminal, or meter box outside the building wall; the terminal or meter box; and the meter.

(Continued on Sheet No. 6.301)

(Continued from Sheet No. 6.300)

SECTION 12.2 GENERAL

12.2.1 Application

This tariff section applies to all requests for underground electric distribution facilities where the facilities requested will be substituted for existing overhead electric distribution facilities. Any person, corporation, or entity capable of complying with the requirements of this tariff may submit a request as follows. Requests shall be in writing and must specify in detail the overhead electric distribution facilities to be converted or the area to be served by underground electric distribution facilities in lieu of presently existing overhead electric distribution facilities serving said area. Upon receipt of a written request, FPL will determine the feasibility of converting the existing facilities, any necessary revisions to this written request, and the non-refundable deposit amount necessary to secure a binding cost estimate and notify the applicant of said amount.

12.2.2 Contribution-in-Aid-Of-Construction (CIAC)

Upon the payment of a non-refundable deposit by an Applicant, FPL shall prepare a binding cost estimate specifying the contribution in aid of construction (CIAC) required for the installation of the requested underground distribution facilities, where the installation of such facilities is feasible, and provide said estimate to the Applicant upon completion of the estimate along with either an Underground Facilities Conversion Agreement or an Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver. The CIAC amount to be collected pursuant to a binding cost estimate from an Applicant shall not be increased by more than 10 percent of the binding cost estimate to account for actual costs incurred in excess of the binding cost estimate. However, the CIAC may be subject to increase or refund if the project scope is enlarged or reduced at the request of the Applicant, or the CIAC is found to have a material error prior to the commencement of construction. The binding cost estimate provided to an Applicant shall be considered expired if the Applicant does not enter into either an Underground Facilities Conversion Agreement or an Underground Facilities Conversion Agreement or an Underground Facilities within 180 days of delivery of the binding cost estimate to the Applicant by FPL.

(Continued on Sheet No. 6.310)

(Continued from Sheet No. 6.301)

12.2.3 <u>Non-Refundable Deposits</u>

The non-refundable deposit for a binding cost estimate for conversion to a direct buried cable in conduit underground electric distribution system shall be determined by multiplying the number of pole line feet of existing overhead electric distribution facilities to be converted by \$1.20. The deposit must be paid to FPL to initiate the estimating process. The deposit will not be refundable, however, it will be applied in the calculation of the CIAC required for the installation of underground distribution facilities. The deposit and the preparation of a binding cost estimate are a prerequisite to the execution of either an Underground Facilities Conversion Agreement or an Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver. If the request for underground electric distribution facilities involves the conversion of less than 250 pole line feet of existing overhead facilities, then no deposit will be required for a binding cost estimate, provided, however, that all other requirements of this tariff shall still apply.

12.2.4 Non-Binding Cost Estimates

Any person, corporation, or entity may request a non-binding cost estimate free of charge. The non-binding cost estimate shall be an order of magnitude estimate to assist the requestor in determining whether to go forward with a binding cost estimate. Neither an Underground Facilities Conversion Agreement nor an Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver may be executed on the basis of a non-binding cost estimate.

12.2.5 Underground Facilities Conversion Agreement

Any Applicant seeking the installation of underground distribution facilities pursuant to a written request hereunder shall execute either the Underground Facilities Conversion Agreement set forth in this tariff at Sheet No. 9.720 or, if applicable, the Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver set forth in this tariff at Sheet No. 9.725. The applicable Agreement must be executed and the CIAC paid by the Applicant within 180 days of the delivery of the binding cost estimate to the Applicant. Failure to execute the applicable Agreement and pay the CIAC specified in the Agreement within the 180 day time limit, or termination of the Agreement, shall result in the expiration of the binding cost estimate. Any subsequent request for underground facilities will require the payment of a new deposit and the presentation of a new binding cost estimate. For good cause FPL may extend the 180 day time limit. Upon execution of either the Underground Facilities Conversion Agreement or the Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver, payment in full of the CIAC specified in the binding cost estimate, and compliance with the requirements of this tariff, FPL shall proceed to convert the facilities identified in a timely manner. However, new service extensions, maintenance and reliability projects, and service restorations shall take precedence over facilities conversions.

12.2.6 Simultaneous Conversion of Other Pole Licensees

Before the initiation of any project to provide underground electric distribution facilities pursuant to either an Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver the Applicant shall have executed agreements with all affected pole licensees (e.g. telephone, cable TV, etc.) for the simultaneous conversion of those pole licensees' facilities and provide FPL with an executed copy of the Agreement(s). Such agreements shall specifically acknowledge that the affected pole licensees will coordinate their conversion with FPL and other licensees in a timely manner so as to not create unnecessary delays. Failure to present FPL with executed copies of any necessary agreements with affected pole licensees within 180 days after delivery of the binding cost estimate to the Applicant shall result in the expiration of the binding cost estimate, the return of any CIAC paid, and the termination of any Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver entered into between the Applicant and FPL.

12.2.7 Easements

Before the initiation of any project to provide underground electric distribution facilities pursuant to either an Underground Facilities Conversion Agreement or an Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver, the Applicant shall provide FPL, at no cost to FPL, all easements, including legal descriptions of such easements and all survey work associated with producing legal descriptions of such easements, specified as necessary by FPL to accommodate the requested underground facilities along with an opinion of title that the easements are valid. Failure to provide the easements in the manner set forth above within 180 days after the delivery of the binding cost estimate to the Applicant shall result in the expiration of the binding cost estimate, the return of any CIAC paid, and the termination of any Underground Facilities Conversion Agreement or Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver entered into between the Applicant and FPL.

(Continued on Sheet No. 6.320)

(Continued from Sheet No. 6.310)

12.2.8 Affected Customer Services

The Applicant shall be responsible for the costs associated with any modifications to the service facilities of customers affected by the conversion of FPL distribution facilities which are made necessary as a result of the conversion. The Applicant shall be responsible for arranging the conversion of affected residential overhead customer service facilities by providing, at no cost to FPL:

- a) any necessary rearranging of the customer's existing electric service entrance facilities to accommodate an underground service lateral through the use of a licensed electrical contractor, in accordance with all local ordinances, codes, and FPL specifications; and
- b) a suitable trench, install FPL provided conduit according to FPL specifications to a point designated by FPL, and perform the backfilling and any landscape, pavement or other similar repairs

FPL shall be responsible for the installation of the service lateral cable, the cost of which shall be included in the Applicant's binding cost estimate. In the event a customer does not allow the Applicant to convert the customer's affected overhead services, or the Applicant fails to comply with the above requirements in a timely manner consistent with FPL's conversion construction schedule, then the Applicant shall pay FPL, in addition to the CIAC specified in the binding cost estimate, the costs associated with maintaining service to said customer through an overhead service drop. The cost for maintaining an overhead service drop from an underground system shall be:

- a) the sum of \$789 for residential dwellings containing less than five individual units; or,
- b) the estimated cost to maintain service for residential dwellings containing five or more individual units.

For existing residential underground service laterals affected by a conversion the Applicant shall be responsible for the trenching, backfilling and any landscape, pavement or other similar repairs and installation of FPL provided conduit, according to FPL specifications, necessary to bring existing underground service laterals of affected customers to an FPL designated handhole or transformer. FPL will install the necessary cable, the cost of which shall be included in the binding cost estimate. However, in the event that a customer owned service lateral fails on connection to the underground distribution system the customer will be responsible for the replacement of their service lateral or compliance with section 10.5 of FPL's tariff.

The Applicant's responsibilities for modifications to the service facilities of non-residential customers affected by the conversion of FPL distribution facilities which are made necessary as a result of the conversion will be specified in an attachment to any Underground Facilities Conversion Agreement or Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver.

12.2.9 Other Terms and Conditions

Through the execution of either the Underground Facilities Conversion Agreement set forth in this tariff at Sheet No. 9.720 or the Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver set forth in this tariff at Sheet No. 9.725 the Applicant agrees to the following:

- a) The Applicant shall be responsible for all restoration of, repair of, or compensation for, property affected, damaged, or destroyed, to accommodate the installation of underground distribution facilities and the removal of FPL's overhead distribution facilities;
- b) subject to section 2.7 Indemnity to Company, or section 2.71 Indemnity to Company Governmental, FPL's General Rules and Regulations, the Applicant shall indemnify FPL from any claim, suit, or other proceeding, which seeks the restoration of, or repair of, or compensation for, property affected, damaged, or destroyed, to remove existing facilities or to accommodate the installation of underground distribution facilities arising from or brought as a result of the installation of underground distribution facilities;
- c) the Applicant shall clear easements provided to FPL of trees, tree stumps and other obstructions that conflict with construction or installation of underground distribution facilities in a timely manner consistent with FPL's construction schedule.

(Continued on Sheet No. 6.330)

(Continued from Sheet No. 6.320)

12.2.10 Type of System Provided

An underground distribution system will be provided in accordance with FPL's current design and construction standards.

12.2.11 Design and Ownership

FPL will design, install, own, and maintain the electric distribution facilities up to the designated point of delivery except as otherwise noted. The Applicant may, subject to a contractual agreement with FPL, construct and install all or a portion of the underground distribution facilities provided that:

- a) such work meets FPL's construction standards;
- b) FPL will own and maintain the completed distribution facilities;
- c) the construction and installation of underground distribution facilities by the Applicant is not expected to cause the general body of ratepayers to incur greater costs;
- d) the Applicant agrees to pay FPL's current applicable hourly rate for engineering personnel for all time spent for (i) reviewing and inspecting the Applicant's work done, and (ii) developing any separate cost estimate(s) that are either requested by the Applicant to reflect only FPL's portion of the work or are required by FPL to reflect both the Applicant's and FPL's portions of the work for the purpose of a GAF Waiver calculation pursuant to an Underground Facilities Conversion Agreement Governmental Adjustment Factor Waiver; and
- e) the Applicant agrees to rectify any deficiencies found by FPL prior to the connection of any Customers to the underground electric distribution system and the removal of the overhead electric distribution facilities.

12.2.12 Relocation

Where underground electric facilities are requested as part of, or for the purpose of, relocation, the requirements of this tariff shall apply. As applicable, the Underground Facilities Conversion Agreement or the Underground Facilities Conversion Agreement - Governmental Adjustment Factor Waiver shall be executed as an addendum to the relocation agreement between FPL and the Applicant. In the event of any conflict between the relocation agreement and this tariff, the tariff shall control. Furthermore, where the regulations of the Federal or State Department of Transportation (DOT) prevent pre-payment of deposits and other conversion costs, the Federal or State DOT may pay the CIAC after the work has been performed.

FLORIDA POWER & LIGHT COMPANY

SUPPLEMENT TO GENERAL RULES AND REGULATIONS FOR THE INSTALLATION OF UNDERGROUND ELECTRIC DISTRIBUTION FACILITIES TO SERVE SMALL COMMERCIAL/INDUSTRIAL CUSTOMERS

SECTION 13.1 DEFINITIONS

The following words and terms, when used in Section 13 shall have the meaning indicated:

<u>APPLICANT</u> - Any person, partnership, association, corporation, or governmental agency that applies for the installation of underground distribution facilities to serve the electrical requirements of a new commercial/industrial building.

BUILDING - Any structure designed for commercial/industrial application.

<u>CABLE IN CONDUIT SYSTEM</u> - Underground distribution system where all underground primary, secondary, service and street light conductors are installed in direct buried conduit. Other facilities associated with cable in conduit, such as transformers, may be above ground.

<u>COMMISSION</u> - The Florida Public Service Commission.

<u>COMPANY</u> - The Florida Power & Light Company. (FPL)

<u>DISTRIBUTION SYSTEM</u> - Electric service facilities consisting of primary and secondary conductors, service laterals, conduits, transformers, and necessary accessories and appurtenances for the furnishing of electric power at utilization voltage.

FEEDER MAIN – A three-phase primary installation, including switches, which serves as a source for primary laterals and loops through suitable overcurrent devices.

FINAL GRADE - The ultimate elevation of the ground, paved or unpaved, which will prevail in a tract of land.

LOOP - An Underground Primary Lateral having two sources of feed at the primary level.

OVERHEAD SYSTEM - Distribution system consisting of primary, secondary and service conductors and aerial transformers supported by poles.

POINT OF DELIVERY - The point where the Company's wires or apparatus are connected to those of the Customer. See Section 13.2.10.

<u>PRIMARY LATERAL</u> - That part of the electric distribution system whose function is to conduct electricity at the primary level from the feeder main to the transformers serving the secondary street mains. It usually consists of one, two or three conductors of insulated cable in conduit, together with necessary accessory equipment for supporting, terminating and disconnecting from the primary mains by a fusible element.

RADIAL- An Underground Primary Lateral having one source of feed at the primary level.

<u>UNDERGROUND SERVICE FACILITIES</u> - The entire length of underground service conductors and associated equipment from the Applicant's property line to the designated point of delivery.

SECTION 13.2 UNDERGROUND DISTRIBUTION FACILITIES TO SMALL COMMERCIAL/INDUSTRIAL CUSTOMERS

13.2.1 Application

This tariff section applies to all requests for Underground Service Facilities made by small commercial/industrial Applicants for new service as is specified below:

- a) Must be a new commercial/industrial installation served by transformer sizes of 100 KVA or less for single or two phase and 300 KVA or less for three phase; and
- b) Must be installed on the Applicant's property beginning at a point along the Applicant's property line and terminating at the Company's designated point of delivery.

The application of this tariff is in addition to and supplements the Company's other rules regarding extensions of facilities for service. An additional contribution-in-aid-of-construction may be required by those rules for extensions or installations of facilities necessary to accommodate a request for Underground Service Facilities made under this section.

13.2.2 Early Notification and Coordination

In order for the Company to provide service when required, it is necessary that the Applicant notify the Company during the early stages of planning projects. Close coordination is necessary throughout the planning and construction stages by the Company, the architect, the builder, the subcontractors and the consulting engineer to avoid delays and additional expense. Particular attention must be given to the scheduling of the construction of paved areas and the various subgrade installations of the several utilities. Failure of the Applicant to provide such notification and coordination shall result in the Applicant paying any additional costs incurred by the Company as a result of said failure.

13.2.3 Changes to Plans, Layout or Grade

The Applicant shall pay for any additional costs imposed on the Company by Applicant due to changes made in the development layout or final grade subsequent to an agreement. These costs include, but are not limited to, engineering design, administration and relocation expenses.

13.2.4 <u>Type of System Provided</u>

The costs quoted in these rules are for underground distribution primary/secondary conductors in direct buried conduit with above-grade appurtenances of standard Company design, excluding throwover service. Throwover service availability and its cost are determined by the Company on an individual basis. Unless otherwise stated, service will be provided at single or two-phase 120/240 volts or, where available, three phase 120/208 volts or 277/480 volts.

13.2.5 Design and Ownership

The Company will design, install, own and maintain the electric distribution facilities up to the designated point of delivery except as otherwise noted. Any payment made by the Applicant under the provisions of these Rules will not convey to the Applicant any rights of ownership or right to specify Company facilities utilized to provide service.

(Continued on Sheet No. 6.510)

(Continued from Sheet No. 6.500)

13.2.6 <u>Rights of Way and Easements</u>

The Applicant shall record and furnish satisfactory rights of way and easements, including legal descriptions of such easements and all survey work associated with producing legal descriptions of such easements, as required by and at no cost to the Company prior to the Company initiating construction. Before the Company will start construction, these rights of way and easements must be cleared by the Applicant of trees, tree stumps and other obstructions that conflict with construction, staked to show property corners and survey control points, and graded to within six inches of final grade, with soil stabilized. In addition, the Applicant shall provide stakes showing final grade along the easement. Such clearing and grading must be maintained by the Applicant during construction by the utility.

13.2.7 <u>Contribution and Credits</u>

The Applicant shall pay the required contribution upon receipt of written notification from the Company. No utility construction shall commence prior to execution of the Underground Distribution Facilities Installation Agreement set forth in Tariff Sheet Nos. 9.700, 9.701 and 9.702 and payment in full of the entire contribution. Where, by mutual agreement, the Applicant performs any of the work normally performed by the Company, the Applicant shall receive a credit for such work in accordance with the credit amounts contained herein, provided that the work is in accordance with Company specifications. Such credits shall not exceed the total differential costs. The credit will be granted after the work has been inspected by the Company and, in the case of Applicant-installed conduit, after the Company pulls all applicable conductors.

13.2.8 Location of Distribution Facilities

Underground distribution facilities will be located, as determined by the Company, to maximize their accessibility for maintenance and operation. The Applicant shall provide accessible locations for meters and transformers when the design of a commercial/industrial building or its appurtenances limit perpetual accessibility for reading, testing, or making necessary repairs and adjustments.

13.2.9 Special Conditions

The costs quoted in these rules are based on conditions which permit employment of rapid construction techniques. The Applicant shall be responsible for necessary additional hand digging expenses other than what is normally provided by the Company. The Applicant is responsible for clearing, compacting, stump removal, paving, and addressing other special conditions. Should paving, grass, landscaping or sprinkler systems be installed prior to the construction of the underground distribution facilities, the Applicant shall pay the added costs of trenching and backfilling and be responsible for restoration of property damaged to accommodate the installation of underground facilities.

13.2.10 Point of Delivery

The point of delivery shall be determined by the Company, but normally will be at or near the part of the building nearest the point at which the Company's electric supply is available to the property. When a location for a point of delivery different from that designated by the Company is requested by the Applicant and approved by the Company, the Applicant shall pay the estimated full cost of the primary/secondary lateral length, including labor and materials, required in excess of that which would have been needed to reach the Company's designated point of delivery. Any redesignation requested by the Applicant shall conform to good safety and construction practices as determined by the Company. Laterals shall be installed, where possible, in a direct line to the point of delivery.

13.2.11 Location of Meter and Raceway

The Applicant shall install a meter trough at the point designated by the Company and a raceway to accept the service lateral conductors if needed. Both will be installed in accordance with the Company's specifications.

(Continued on Sheet No. 6.520)

(Continued from Sheet No. 6.510)

13.2.12 Contribution by Applicant

The Applicant shall pay the Company the average differential cost between installing overhead and underground distribution facilities based on the following:

a) Primary lateral, riser (if from overhead termination point), pad mounted transformer and trench with cable-in-conduit not to exceed 150 feet in radials and 300 feet in loops.

Applicant's Contribution	
	From Existing
From Overhead	Underground
Termination Point	Termination Point
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
\$ 0.00	\$ 0.00
	From Overhead <u>Termination Point</u> \$ 0.00 \$ 0.00

b) Secondary riser and lateral, excluding handhole or junction box, with connection to Applicant's service cables no greater than 20 feet from Company riser pole.

1) Small single phase	\$ 601.33
2) Large single phase	\$ 1,085.49
3) Small three phase	\$ 884.63
4) Large three phase	\$ 1,609.40

c) FPL service cable installed in customer provided and customer installed 2" PVC (for main line switch size limited to 60 amps for 120V, 2 wire service, or 125 amps for 120/240v, 3 wire service) where customer's meter can is at least 5 feet and no more than 100 feet from the FPL pole.

	120v 60 amp	120/240v 125 amp
	2 wire service	3 wire service
1) Installed on a wood pole - accessible locations	\$ 506.26	\$ 463.18
2) Installed on a wood pole - inaccessible locations	\$ 584.61	\$ 528.23
3) Installed on a concrete pole - accessible locations	\$ 569.74	\$ 526.65

d) Handholes and Padmounted Secondary Junction Box, excluding connections.

1) Handhole	
a. Small - per handhole	\$232.68
b. Intermediate - per handhole	\$286.94
c. Large - per handhole	\$533.21
2) Pad Mounted secondary Junction Box - per box	\$3226.71

3) Pad Mounted secondary Junction Cabinet, used when electrical loads exceed the capacity of the secondary junction box (above) or when the number of the service conductors exceed the capacity of the pad mounted transformer. This charge is only applicable if the majority of the customer's service conductor diameter is less than 500 MCM.

Per cabinet (includes connecting up to 12 sets of conductor)	\$11,	704.68
Tapping service conductors (if more than 12 sets) – per set	\$	88.00

(Continued on Sheet No. 6.530)

(Continued from Sheet No. 6.520)

e) Primary splice box including splices and cable pulling set-up.

1) Single Phase - per box	\$1,109.75
2) Two Phase - per box	\$1,660.91
3) Three Phase - per box	\$1,867.45

f) Additional installation charge for underground primary laterals including trench and cable-in-conduit which exceed the limits set in 13.2.12 a).

1) Single Phase - per foot	\$ 0.98
2) Two Phase - per foot	\$ 3.02
3) Three Phase - per foot	\$ 1.81

g) Additional installation charge for underground primary laterals including trench and cable-in-conduit extended beyond the Company designated point of delivery to a remote point of delivery.

1) Single Phase - per foot	\$ 9.41
2) Two Phase - per foot	\$ 13.88
3) Three Phase - per foot	\$ 15.29

h) The above costs are based upon arrangements that will permit serving the local underground distribution system within the commercial/industrial development from overhead feeder mains. If feeder mains within the commercial/industrial development are deemed necessary by the company to provide and/or maintain adequate service and are required by the Applicant or a governmental agency to be installed underground, the Applicant shall pay the company the average differential cost between such underground feeder mains within the commercial/industrial development and equivalent overhead feeder mains, as follows:

Applicant's

	Co	ntribution
Cost per foot of feeder trench within the commercial/industrial		
development (excluding switches)	\$	10.09
Cost per above ground padmounted switch package	\$25	5,716.84

 The Company will provide one standby/assistance appointment at no additional charge to the Applicant adding new or additional load to assist with installation of the Applicant's conductors and conduit(s) into a padmounted transformer, pedestal or vault (not to exceed four hours in duration) during normal hours of operation. Additional appointments will be provided upon request, at the Applicant's expense.

(Continued on Sheet 6.540)

(Continued from Sheet No. 6.530)						
13.2.13	Cor	ntribution Adjustments				
	a)	Credits will be allowed to the Applicant's contribution in Section provides trenching and backfilling for the Company's facilities.	C A	2.2.12. where, by mutual agreement, the Applicant redit to the Applicant's ontribution		
		 Credit per foot of primary trench Credit per foot of secondary trench 	\$ \$	4.07 3.23		
	b)	Credits will be allowed to the Applicant's contribution in section installs Company-provided conduit per Company instructions.	13.	.2.12. where, by mutual agreement, the Applicant		
		 Credit per foot of 2" conduit Credit per foot of larger than 2" conduit 	\$ \$	0.70 0.98		
	c)	Credit will be allowed to the Applicant's contribution in Section installs a Company-provided handhole per Company instructions,	13.	2.12. where, by mutual agreement, the Applicant		
		 Credit per large handhole/primary splice box Credit per small handhole 		272.05 71.52		
	d)	Credit will be allowed to the Applicant's contribution in Section installs a Company-provided concrete pad for a pad-mounted tran- instructions,				
		Credit per pad	\$	70.12		
	e) Credit will be allowed to the Applicant's contribution in Section 13.2.12. where, by mutual agreement, the Applicant installs Company-provided concrete pad for a pad-mounted feeder switch chamber per Company instructions,					
		Credit per pad	\$	660.48		
	f) Credit will be allowed to the Applicant's contribution in Section 13.2.12. where, by mutual agreement, the Applicant installs Company-provided concrete pad for a feeder splice box per Company instructions,					
		Credit per splice box	\$	776.87		

RESERVED FOR FUTURE USE

.

RESERVED FOR FUTURE USE

RESERVED FOR FUTURE USE